REVIEWS OF TOPICAL PROBLEMS

Contents

PACS numbers: 61.43.Fs, 64.70.kj, 64.70.Q-

Modern aspects of the kinetic theory of glass transition

T V Tropin, J W P Schmelzer, V L Aksenov

1.	Introduction	42
2.	Experimental observations	43
3.	Theoretical methods	47
	3.1 Fictive temperature concept; 3.2 Vol'kenshtein-Ptitsyn relaxation theory; 3.3 Free-volume models; 3.4 Tool-	
	Narayanaswamy-Moynihan method; 3.5 Kovacs method; 3.6 Adam-Gibbs theory; 3.7 Configurational entropy	
	approach to glass transition; 3.8 Glass transition in terms of nonequilibrium thermodynamics; 3.9 Some current	
	microscopic approaches to glass transition description	
4.	Kinetic criteria of glass transitions	57
	4.1 General kinetic criterion of glass transition and examples of its application; 4.2 Width of a glass transition range	
5.	Physical relaxation of polymers	60
6.	Glass transition kinetics in a wide range of cooling rates	61
7.	Conclusions	62
	References	64

<u>Abstract.</u> This paper reviews glass transition kinetics models that are developed to describe the formation of structural (for example, covalent and metallic) glasses, as well as to account for the transition of a polymer to a solid glassy state. As the two approaches most frequently used over the last decade to model the glass transition, the Tool–Narayanaswamy–Moynihan model and the Adam–Gibbs theory of glass transition are described together with examples of their applications. Also discussed are entropy-based approaches that rely on irreversible thermodynamics methods originated in the work of De Donder, Mandelstam, and Leontovich. The actual problems that arise in applying these methods and the prospects of their development are discussed. A brief overview of statistical glass transition models is given, including the mode-coupling and energy-landscape theories.

T V Tropin Frank Laboratory of Neutron Physics, Joint Institute for Nuclear Research, ul. Joliot-Curie 6, 141980 Dubna, Moscow region, Russian Federation E-mail: ttv@jinr.ru J W P Schmelzer Institute of Physics, University of Rostock, Wismarsche Str. 43-45, 18057 Rostock, Germany; Bogoliubov Laboratory of Theoretical Physics, Joint Institute for Nuclear Research, ul. Joliot-Curie 6, 141980 Dubna, Moscow region, Russian Federation E-mail: juern-w.schmelzer@uni-rostock.de V L Aksenov National Research Centre 'Kurchatov Institute', Petersburg Nuclear Physics Institute, Orlova roshcha, 188300 Gatchina, Leningrad region, Russian Federation; Frank Laboratory of Neutron Physics, Joint Institute for Nuclear Research, ul. Joliot-Curie 6, 141980 Dubna, Moscow region, Russian Federation E-mail: dir@pnpi.spb.ru Received 17 July 2015, revised 8 October 2015 Uspekhi Fizicheskikh Nauk 186 (1) 47-73 (2016)

DOI: 10.3367/UFNr.0186.201601c.0047 Translated by E G Strel'chenko; edited by A Radzig **Keywords:** glass transition, kinetics of glass transition, polymer glass transition, theory and modeling of glass transition, thermal properties of amorphous solids and glasses

1. Introduction

The glass transition, the dynamical transition of a system from a thermodynamically equilibrium liquid state to a frozen-in thermodynamically nonequilibrium (glassy) state, can be regarded as one of the earliest technologies humans tapped to fabricate materials [1–4]. With new materials being invoked and increasingly higher variation rates of temperature and other external parameters becoming available, new glass fabrication methods are now emerging and their range of application continuously expands. One finds numerous examples of how solid amorphous materials produced in the glass transition are used in every sphere of human activity and industry. One such example is polymers, namely materials which have a wide range of applications and show enormous promise for the future.

The fact that the final state of a material is strongly dependent on its cooling rate and thermal prehistory imparts great applied importance to the problem of the theoretical description of the glass transition process. Considerable optimization can be achieved for glass-based production processes and for the performance of glass-based products with a theory capable to quantitatively account for glass transition in various materials. However, neither the glass structure itself nor the glass formation kinetics have yet been given a sufficient explanation at the microscopic level and thus they represent an intriguing problem for condensed matter physics.

Although low-molecular materials, polymers, metallic glasses, and other glass-forming systems differ considerably in their molecular structure whether in a liquid or a solid amorphous state, there is much observational evidence that, when changed to glass, their macroscopic characteristics are essentially similar. For example, the specific volume, thermal expansion coefficient, enthalpy, elastic modulus, and other material properties change in a qualitatively identical manner in completely different materials, irrespective of their molecular structure, suggesting that a general theory should exist that is at least capable of predicting how their macroscopic properties evolve.

The challenge of creating such a general theory is well illustrated by the fact of existing various and diverse phenomenological approaches to and theories for describing the glass transition process that in some cases differ fundamentally in some aspects and which are too numerous to describe in detail within a single review. This diversity itself implies, however, that the theory of glass transition currently represents one of the most significant unsolved topics in theoretical physics. The important point here is that, in addition to the well-known problems of describing real equilibrium systems by statistical physics methods, one faces problems related to the change in and the rate of change of the external parameters involved. Thus, opinions still vary as to whether glass transition is a purely dynamic process which 'freezes' relaxation phenomena characteristic of liquids or whether, at least under some conditions, it results from the existence of a special thermodynamic phase, for example, socalled ideal glass. A further point to note is that, as tools for describing experimental data, theoretical methods formulated over fifty years ago still remain popular. However, the last few decades have seen active attempts to develop new microscopic approaches closely relying on the laws of statistical physics and, hence, on the employment of computer simulations capable of implementing various (including ab initio) calculation methods.

This review systematically summarizes the approaches that are currently most widely applied for describing the glass transition kinetics. Because, as mentioned, the glass transition process is similar for completely different materials, the theoretical methods are usually described here without regard for the structural features or properties of a specific material. The examples chosen and references cited mainly reflect the author's research interests, which in recent years have concerned the description of experimental data on polymers. This approach does not lead to a limitation of the range of application of the methods to be described; they are quite successful in describing the glass transition in any material. Furthermore, the methods to be discussed are also of interest for describing glasslike states that arise at phase transitions, such as structural transitions [5, 6] and superconducting transitions occurring in high-temperature superconductor glasses [7-9]. Mention should also be made of similar states observed in disordered crystals [1].

On the other hand, the methods developed in the theory of phase transitions (for example, in the theory of spin glasses) are also utilized in the general theories of glass transition, and similar ideas appear in related fields such as the study of metallic [10], dipole [11], and polymer [12] glasses.

The particular emphasis of this review is on the application of established phenomenological models to the theoretical methods under discussion because, despite their high promise, current microscopic theories cannot yet quantitatively describe the glass transition kinetics of complex systems. Moreover, such phenomenological approaches provide a general picture which, as is the case in classical thermodynamics, can be developed further through the employment of microscopic methods.

The review outline is as follows. Section 2 briefly presents what is typically observed experimentally in glass transitions. Section 3 supplies step by step a chronological account of the theoretical methods that have been developed for describing glass transition kinetics, starting from Tool's concept of fictive temperature and ending with the mosaic and modecoupling theories. Section 4 presents the generalized kinetic glass transition criterion and discusses the general implications of the criterion and its application to the study of the effect of cooling/heating rate on the glass transition temperature and the width of transition interval. Section 5 briefly discusses recent advances in the theoretical description of the physical relaxation of polymers. Section 6 presents the results of modeling of the glass transition of polymers in a wide range of cooling/heating rates and discusses the challenges that exist in this field. The concluding section discusses the results of some studies, examines topical problems in glass formation kinetics, and explores prospects of theoretical approaches to solving these problems.

2. Experimental observations

The way in which a diversity of properties of a system (the specific volume, enthalpy, etc.) evolve in time as a glass transition proceeds can give a clue to the basic features of the process. This is clearly exemplified by the behavior of a certain property p for two temperature 'histories' which qualitatively correspond to most experimental situations: (1) a sudden switch of temperature, when it abruptly changes by ΔT throughout the system, and (2) a temperature decrease at a certain (usually constant) rate q = dT/dt. For a stepwise isobaric change in temperature, the glass transition region reveals two contributions to the change in p: one of them instantaneous (isostructural), and the other relaxational (structural) (Fig. 1a). The former type of response (which remains relevant for a glassy state as well) reflects how the thermal vibrations of the material's particles (phonon contribution) change their nature as a result of the change in the external conditions, whereas the latter response mechanism is due to a change in the structural characteristics of the system and remains inoperative in the 'frozen-in state' of the material.

An important feature distinguishing glass transition relaxation from relaxation processes in many other systems is that, for structural relaxation, the chosen parameter pvaries *nonexponentially* with time t. Whereas relaxation in a liquid at sufficiently high temperatures follows the law $p(t) \sim \exp(-t/\tau)$ (where τ is the characteristic relaxation time), in the glass transition region we observe a strong deviation from this law, with different properties of the system possibly demonstrating different behaviors [13, 14]. The response function in this case is most commonly—and quite reasonably-approximated by the so-called Kohlrausch law, $p(t) \sim \exp\left[-(t/\tau)^{\beta}\right]$ [15, 16], where β is a positive factor less than unity, usually $0.35 < \beta < 0.75$ [2]. The case of β equal to unity corresponds to Maxwell's exponential relaxation behavior, whereas $\beta < 1$ results in what is known as stretched exponential relaxation. Another feature of this relaxation is its nonlinear nature, which manifests itself in the relaxation time τ depending on both the conventional thermodynamic parameters (temperature, pressure, glass composition) and additional structural parameters of the

Figure 1. (Color online.) A characteristic response of a glass-forming system expressed in terms of the dependence of p on temperature or time (a) for a jump in temperature, and (b) for a liquid-to-glassy state transition at a constant cooling rate (blue lines) and a subsequent heating (red lines). Here, p_{eq} , p_{eq}^0 , and p_{eq}^1 are the equilibrium values of the property p for different temperatures; q_c is the heating/cooling rate, which assumes three values; T_g^1 , T_g^2 , and T_g^3 are the glass transition temperatures for the respective values of the rate q_c , and p_g^1 , p_g^2 , and p_g^3 are the corresponding values of the property p which are 'frozen-in' during the glass transition.

glass that appear in a thermodynamic description due to the departure of the glassy state from equilibrium [1-3].

Figure 1b illustrates how a system's property p being studied (say, enthalpy, specific volume, etc.) evolves as the system is uniformly cooled at different constant rates from a liquid state [17, 18]. As the temperature decreases, the relaxation time of the system increases, resulting in the deviation of the observed values of p from the equilibrium value p_e in a certain range of temperature T, so that p assumes a certain 'frozen-in' value p_g on the time interval of experimental observations. Note that the higher the cooling rate q, the larger the 'frozen-in' value p_g . If upon completion of cooling the temperature is steadily increased (a glass-toliquid transition), then the behavior of p in the glass transition region will differ from its behavior during cooling, giving rise to hysteresis. There is a wealth of experimental evidence that supports this result and which also shows that the glass transition temperature depends on the heating/cooling rate. Other material properties, such as the isobaric heat capacity C_p , vary with temperature in a different way (as the derivative

with respect to T of the dependences depicted in Fig. 1b), showing a monotonic change in the glass transition region from a linear behavior for a liquid to a different linear behavior for a glassy state. At the reverse transition, a nonmotonic curve with one or two heat capacity peaks is observed if the heating proceeds sufficiently fast (faster than the cooling rate).

One of the key conclusions to draw from the curves plotted in Fig. 1 is that the glass transition is essentially a kinetic process. The evolution of the properties of a system depends on both the rate of change of the external parameters (here, temperature) and the time needed for the system to relax to the corresponding equilibrium state. To describe the kinetic transition to a glassy state, various combinations of these parameters have been proposed, starting from the works by Bartenev, Jones, Ritland, Volkenstein, and Ptitsyn, which have led to the development of a number of criteria (see Refs [14, 19] for a review). As a generalization of the earlier proposed criteria, Refs [14, 19] also formulated a glass transition criterion based entirely on a general thermodynamic analysis of the glass transition, without relying on any specific models of a glass-forming system. This generalized criterion requires that the relaxation time $\tau_{\rm R}$ of the structural parameter of the system be equal to the characteristic time for the variation of the glass transition-inducing external factor. If the transition of the system to a glassy state is due to a change in temperature, then the characteristic time for the variation of temperature τ_T can be written as

$$\tau_T = \left(\frac{1}{T} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right| \right)^{-1}.$$
 (1)

Then, the condition for the transition from an equilibrium liquid state (in which $\tau_{\rm R} \ll \tau_T$) to a 'frozen-in' glassy state $(\tau_{\rm R} \gg \tau_T)$ is as follow: $\tau_{\rm R} \sim \tau_T$ [14, 19], from which the glass transition temperature $T_{\rm g}$ can be estimated as

$$\left. \left(\frac{1}{T} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right| \tau_{\mathrm{R}} \right) \right|_{T=T_{\mathrm{g}}} \approx 1.$$
⁽²⁾

References [14, 19] show that the earlier developed glass transition criteria are limiting cases, or approximations, of general criterion (2). Notice also that if a transition to a glassy state is caused by changes in other external parameters determining the state of the system (for example, pressure), then the generalized criterion remains unchanged in form requiring merely that τ_T is replaced by the characteristic time for the variation of the corresponding parameter (see Section 4.1 for more detail).

Using the glass transition criterion (2) and employing the appropriate expressions for the relaxation time of a specific system, it is possible to obtain analytical expressions for the glass transition temperature as a function of the cooling/ heating rate. This was first done by Bartenev [20], who approximated criterion (2) as $q|\tau_R| \approx \text{const}$, where |q| is the heating (or cooling) rate of the system. Assuming the temperature dependence of the relaxation time to be defined by $\tau_R = \tau_{R0} \exp \left[-A/(k_B T)\right]$, Bartenev obtained the expression, now known as the Bartenev–Ritland equation [20–22], for the dependence of the glass transition temperature on the rate q:

$$\frac{1}{T_{\rm g}} = C_1 - C_2 \ln q \,, \tag{3}$$



where the material parameters C_1 and C_2 are determined from the relaxation behavior parameters. Using general criterion (2), it can be shown that C_1 is not a constant, but rather a function weakly dependent on T_g [which does not usually greatly affect dependence (3)].

There is abundant experimental evidence that the dependence of the glass transition temperature T_g on the heating/ cooling rate is usually described well by formula (3) over a wide rate range. The reason is that, for this quite wide but still limited temperature range, the relaxation time is well described by Maxwell's expression Bartenev used. However, for a wider range of the temperature variation rate, the glass transition temperature T_g varies over a wider range, making it necessary to derive a more accurate expression for the relaxation time [hence, deviations from equation (3)].

The dependence of the structural relaxation time on temperature is reflected in the temperature dependence of the viscosity of a supercooled liquid. It is a long-known fact that the properties of glass-forming systems have qualitatively different temperature dependences. Indeed, glasses are known to be technologically divided into 'short' and 'long' types depending on how fast the viscosity in the region of the glass transition changes with temperature (see, for example, Ref. [1]). To describe these differences in the behavior of glass-forming melts by dependences of the form $\eta(T)$ or $\tau(T)$, Angell [23] rather recently proposed a new parameter, the fragility index, with which to quantify the deviation from the Arrhenius behavior. Materials that exhibit a typical Arrhenius behavior of $\eta(T)$ are classified as strong, and those deviating mostly from it as fragile. For the latter, the viscosity increases sharply in the glass transition region.

Notice, however, that Angell's classification and its consequences only make sense when the glass transition temperature is defined for all liquids in the same way, from the viscosity value of 10^{12} Pa s independent of the heating or cooling conditions (rates), in accord with Tammann's [24] suggestion about the definition of T_g (see Ref. [25] for details). However, as can be seen already from the Bartenev–Ritland equation (3), temperature—and hence viscosity—can, depending on the cooling or heating rate, assume very different values at the glass transition. Therefore, any criterion specifying the value of viscosity at T_g is of limited value as a tool for stimulating the understanding of glasses and glass transitions.

Whatever theoretical arguments and discussions there are on the value of the fragility index for describing the glass transition, experimental studies show that the temperature dependence of the viscosity of supercooled glass-forming liquids generally tends to be well described by the phenomenological Vogel–Fulcher–Tammann (VFT) equation [26–28]

$$\eta(T) = \eta_0 \exp\left(\frac{B}{T - T_0}\right),\tag{4}$$

where η_0 , *B*, and T_0 are material-specific constants. Therefore, the glass transition is appropriately described by applying Eqn (4) as an expression for the characteristic relaxation times with a corresponding factor for the passage from $\eta(T)$ to $\tau(T)$. It can be shown [1, 17, 18] that over a wide range of *q* rates, the Bartenev–Ritland law is satisfied for strong glass-forming systems, and takes a different form for fragile ones (see Section 4). Equation (4) adequately describes the behavior of the socalled equilibrium viscosity of a liquid, which leads to a similar behavior for the equilibrium relaxation time. As mentioned above in connection with Fig. 1, one more important feature of glass transition is the so-called memory of the material. In this effect, both the thermodynamic and kinetic characteristics of the system, including the way it relaxes to equilibrium, depend not only on the current values of properties of the system (such as pressure, temperature, and the molar fractions of the components) but also on its previous thermal history. The memory effect was observed in early experiments for the example of the refraction index of glasses hardened in different ways; for other experiments revealing these effects, see Refs [29–33].

Hence, not only do the usual thermodynamic parameters such as temperature, pressure, and molar fractions have to be taken into account in describing the evolution of the material's properties in the glass transition process, but so does the thermal history of the material. The inclusion of the memory effect is of interest from a theoretical point of view because it turns out to be not sufficient to consider a firstorder differential equation for a single order parameter for a proper description of the glass transition. In order to develop an appropriate theoretical description, different approaches to the problem may be applied: one can either to consider relaxation of several structural parameters by setting up a system of coupled first-order differential equations, or, on the contrary, to develop a theory with a single structural order parameter introducing, in doing so, 'memory' integrals over the thermal history of the process.

There is a wide range of methods for experimental investigations of glass transitions and glass properties. One of the most effective current methods of experimentally exploring the glass transition in various materials is differential scanning calorimetry (DSC). With the development of new experimental techniques, the previously accessible DSC cooling rates of at most a few kelvins per second increased by many orders of magnitude up to 10^6 K s⁻¹ [34–37], thus reaching the lowest limits accessible to computer simulation. Having regard to the way they are used and given the extremely long duration of the experiment (for example, cooling to 100 K takes about three years at 10^{-6} K s⁻¹, and three months at 10^{-5} K s⁻¹), the unique possibility arises of investigating the glass transition over the cooling rate range of more than ten orders of magnitude. An alternative (or supplementary) approach constitutes temperature-modulated calorimetry (TMC) [38-41] for the study of the phenomenon of dynamic glass transition (or TMDSC if different device realizations are employed). An experimental study in this case consists of exposing a system to small-amplitude temperature oscillations at a given frequency $(\delta T(t) \sim \exp(i\omega t))$ and then analyzing the system's response. The experimentally accessible frequency range extends from a few µHz to a few hundred kHz for laser alternating current (AC) calorimetry. It should be noted that measurements in this range of ω and q require the employment of several experimental techniques (and, hence, of various devices), as well as the use of samples widely differing in volume (mass) (see, for example, Refs [42, 43]).

Temperature oscillations in the present-day devices are usually accompanied by the cooling or heating of the sample at a preassigned constant rate. In some cases, two glass transition temperatures may be recognized, one of which corresponds to the transition described above, and the other,



Figure 2. Temperature dependence of the specific heat capacity of polystyrene as measured by TMDSC [44].

the dynamic one, to the retardation in the relaxation to equilibrium of some 'internal' system's modes with respect to fast temperature oscillations [44, 45]. The former glass transition temperature depends on the cooling rate, $T_g(q)$, and the second on the temperature oscillation frequency, $T_g(\omega)$. Figure 2 presents the results of such measurements on polystyrene. References [46–48] obtained expressions relating q and ω for the observed transitions, and Refs [42, 49, 50] showed good agreement with experimental data over a wide frequency range for different samples. The reader is referred to handbook [51] for a review including a description of the basics of TMDSC technique and its application to polymer studies.

The schematic model of the dynamic glass transition is similar to Herzfeld's and Rice's model of sound absorption in polyatomic gases [52], which introduces a relaxation time accounting for energy transfer from the system's internal to external degrees of freedom. In this case, one can introduce a complex-valued dielectric constant, with its real and imaginary parts determined experimentally. Similarly, for a model with a constant energy transfer relaxation time, the frequency dependence of C_p is found to be given by [53–55]

$$C_{p}^{*}(\omega) = C_{p}' - iC_{p}'' = C_{p,\infty} + \frac{C_{p}^{i}}{1 + \omega^{2}\tau^{2}} - i\frac{C_{p}^{i}\omega\tau}{1 + \omega^{2}\tau^{2}}.$$
 (5)

From Eqn (5) it is seen that at a certain value of temperature oscillation frequency ω the imaginary part of $C_p(\omega)$ reaches a maximum, and just this value is associated with the dynamic glass transition. Situations that occur in real systems are more complex and require including a distribution of relaxation times and considering how this distribution varies with temperature.

Much of the experimental and theoretical work has been focused on the structural relaxation ('aging') of a material held at temperatures $T_a < T_g$ [56–60], where T_a is located within the glass transition range. This holding is followed by further cooling to a preassigned temperature beyond the glass transition range. For a number of polymers, such as polystyrene, polymethacrylate (PMA), and polymethyl methacrylate (PMMA), measured data are available for both short-time relaxation (from 0.01 s) and annealing for one year and longer [58, 61–65]. The results are presented in the form of the time dependence of the enthalpy regained by the system during its annealing (as opposed to that for a sample cooled without annealing).



Figure 3. DSC observations of two-stage relaxation in polystyrene [65]. $\Delta H(t)$ is the restored enthalpy at instant of time *t*, and $\Delta H_{tot}(t)$ is the total enthalpy which the system can restore. Annealing temperatures are shown.

Over the last decade, a discussion in the literature has been aroused as to whether a system, if given enough time, reaches a state with an enthalpy extrapolated from the data for the liquid, i.e., extrapolated to the equilibrium value, or whether, as some theorists have hypothesized, a new metastable limit exists for $T < T_g$ [66]. One fundamentally important recent finding in this area-specifically, in the study of structural relaxation in polystyrene and polycarbonate — is an observation of two-stage relaxation mechanism [63, 65]. From the experimental data shown in Fig. 3, it is seen that the initial relaxation at sufficiently low temperatures T_a is followed by a time plateau, which can be of various durations and which is, in turn, followed by a renewed structural relaxation ultimately bringing the system to equilibrium. These results continue to be discussed, and new experimental data are expected to come very soon.

From the perspective of investigating the behavior of the properties of various systems in the glassy state, it is also of interest to study variations in these properties and the glass transition kinetics under conditions of confining system's dimensions [10, 67]. Such studies are primarily concerned with the glass transition in pores or film-shaped materials [34, 68–71]. There is considerable recent interest in polymer thin films due to the fact that their thickness can be varied virtually continuously from a few micrometers to a few nanometers [68]. One aspect of this interest is the current miniaturization trend in industry (in particular, in microelectronics).

There is a variety of techniques for determining the glass transition temperature of a film, including ellipsometry, dilatometry, X-ray diffractometry, dielectric spectroscopy, and DSC. For substrate-supported polymer thin films, it has been shown by various methods that, as the film thickness is decreased to below 300 Å, the glass transition temperature decreases to within 25 K of the initial, 'bulk' value T_g [72]. A similar effect has been observed in thin films of other polymers [73, 74]. Studies of PMMA revealed the influence of the substrate type (and, hence, of the polymer–substrate interaction) on the dependence of T_g on the film thickness *d* [75, 76]. It is found that the effect depends on the chain length (molar mass) of the polymer: linearly for heavy and non-linearly for light polymers (in the latter case, the dependence is identical to that for a substrate-supported film [77]).

Over the last decade, it has become clear from simultaneous measurements of the molecular dynamics and glass transition temperature of thin polymer films that the reduction in T_g with decreasing *d* has no direct relation to the change in the relaxation time of the polymer [71, 78–80]. It is conjectured that just the geometry on the nanoscale (for example, the film thickness) markedly influences the relaxation time of the sample studied (which affects the obtained values of T_g), whereas the relaxation time determined by molecular dynamics again corresponds to the bulk values [69].

The standard properties and kinetic parameters of glassforming materials are currently measured routinely both for polymers and other materials. As a result, the totality of accumulated data is so vast that already years ago many electronic databases, some structured by system types and some by property types, were developed [81–83]. Mention should also be made of a number of excellent reviews concerned with the glass transition [23, 84–95], in which the observed behavior patterns and theoretical approaches are discussed.

Overall, the experimental observations described in this section adequately reflect the nature of the kinetics of the glass transition considered as a kinetic phenomenon, which makes it possible to proceed to discussing the theoretical foundations of the process.

3. Theoretical methods

The phenomenological description of the kinetics of the glass transition has a history of over 80 years [1-4, 96]. A simplified view of the glass transition as a transition from an equilibrium liquid to a 'frozen-in' state of glass proceeding at a discrete value of temperature T_g was proposed by Simon [97, 98]. In such a treatment, it is conjectured that for $T > T_g$ the system can achieve equilibrium at a rate greatly exceeding the rate of changing the external parameters, whereas for $T < T_{\rm g}$ the system finds itself in a frozen-in nonequilibrium glassy state and, according to experimental observations, is unable to reach the equilibrium state. There exists, however, conclusive experimental evidence that glass transitions actually occur in a certain temperature range $T_g^- < T_g < T_g^+$. These experimental data are consistent with Tammann's [24] description of the glass transition expressed nearly at the same time as Simon developed his ideas. As shown below, the width of the glass transition region increases with increasing the rate of changing temperature. Sections 3.1-3.9 present and discuss the basic features of currently employed methods of description of the glass transition kinetics.

3.1 Fictive temperature concept

The concept of structural (or, alternatively, fictive) temperature $T_{\rm f}$ was introduced by Tool as an additional characterization parameter for the purpose of describing the supercooled liquid–glass transition of a system [99–101]. It is, by definition, the temperature at which "a given system will be in the equilibrium state if heated at a sufficient rate" [100]. Thus, it is assumed that in this state the system has the same structure as the glass under study, but at a temperature $T_{\rm f}$ it is in equilibrium. The rate of the system relaxation to equilibrium in this case is determined by the difference between the system's current temperature T and its fictive temperature $T_{\rm f}$, and also by the kinetic factor which is inversely proportional to viscosity. In Ref. [100], a qualitative (but not quantitative) correspondence is reported to exist between the experimental and model curves for the evolution of some properties of the system, utilizing these concepts.

The concept of fictive temperature is still actively used even today in the analysis of experimental data in some model frameworks. Thus, from DSC data it is possible to determine the limiting fictive temperature $T'_{\rm f}$ (corresponding to the 'frozen-in' value of $T_{\rm f}$) as an integral characteristic of the C_p curve [102]:

$$\int_{T^*}^{T'_{\rm f}} (C_p^{\rm liq} - C_p^{\rm gl}) \,\mathrm{d}T_{\rm f} = \int_{T^*}^{T'} (C_p - C_p^{\rm gl}) \,\mathrm{d}T, \tag{6}$$

where T^* is a certain temperature beyond the glass transition region at which the heat capacity is equal to its equilibrium value $C_p^{\text{liq}}(T)$; T' is a temperature well below the glass transition region, $T' \ll T_g$, and $C_p^{\text{gl}}(T)$ is the extrapolated linear dependence of the glassy state heat capacity. Currently, this definition of T_f' tends to be viewed as referring to the glass transition temperature T_g as measured by DSC [88, 103]. According to models that rely on the relaxation of enthalpy, the evolution of the fictive temperature is directly related to the heat capacity by

$$C_p^{\rm red} = \frac{\mathrm{d}T_{\rm f}}{\mathrm{d}T} \,, \tag{7}$$

where C_p^{red} is the reduced isobaric heat capacity given by

$$C_{p}^{\text{red}}(T) = \frac{C_{p}(T) - C_{p}^{\text{gl}}(T)}{C_{p}^{\text{liq}}(T) - C_{p}^{\text{gl}}(T)}.$$
(8)

It is seen that, with this definition, the heat capacity equals unity for $T \gg T_g$, and zero for $T \ll T_g$. It should be noted that totally different definitions of the structure parameters are possible (see, for example, the earlier work [104] or discussion in Ref. [105], and criticism of the T_f approach in Ref. [106]), but the fictive temperature concept has proved to be a suitable framework for the quantitative description of the glass transition. While the simplified model initially advanced by Tool is no longer utilized to describe experimental data, some of the more sophisticated models still in use and considered in Sections 3.2–3.9 employ it as their basis.

3.2 Vol'kenshtein-Ptitsyn relaxation theory

The theory of the glass transition developed by Vol'kenshtein and Ptitsyn [107, 108] based on the analysis of transitions in the framework of a two-state model provides a means to qualitatively describe both the kinetics of the glass transition process and the general dependence of its features on the heating and cooling rates of the system.

This theory considers the transitions of the kinetic elements of the system between states 1 and 2 with probabilities of ω_{12} and ω_{21} ; the molar fractions of these elements in these states satisfy the equality $v_1 + v_2 = 1$. Then, for v_2 we have the following equation

$$\frac{dv_2}{dt} = -\frac{1}{\tau} (v_2 - v_2^{eq}), \qquad (9)$$

where v_2^{eq} is the equilibrium value of v_2 , for which $dv_2/dt = 0$, and τ is the relaxation time of the system. Both these values are determined by the transition probabilities which, in turn, depend on the temperature and pressure of the system:

$$x = \frac{1}{\omega_{12} + \omega_{21}}, \quad v_2^{\text{eq}} = \frac{\omega_{12}}{\omega_{12} + \omega_{21}}.$$
 (10)

In their examination of the general form of the solution to equation (9) for heating and cooling processes occurring at a constant pressure and for *T*-dependence of v_2^{eq} assumed to be weaker compared to the relaxation time, the authors of Ref. [100] conducted a detailed analysis of how the quantities linear in v_2 and the derivatives dv_2/dt corresponding to the heat capacity vary with temperature. The variations obtained are in qualitative agreement with experimental data in terms of the heat capacity peak and v_2 hysteresis for various heating-to-cooling rate ratios.

Vol'kenshtein and Ptitsyn also arrived at an estimate for the dependence of the glass transition (softening) temperature on the cooling (heating) rate [107]:

$$\frac{\mathrm{d}\tau}{\mathrm{d}T}\Big|_{T_{\mathrm{g}}} \sim -\frac{1}{q(T_{\mathrm{g}})} \,. \tag{11}$$

Expression (11) provided a correct description (at least qualitatively) of the then available experimental data on the glass transitions in liquids and polymers (see, for example, Refs [20, 21, 109]). Meanwhile, such a model, with τ considered as a function of temperature (pressure) alone, fails to account for some fundamental properties of the glass transition process, one example being the memory effect. Despite their simplicity, the key aspects of Vol'kenshtein's and Ptitsyn's theory are used, variously modified, even in its simplest version in current approaches to the description of glass transition. A few years later, Cooper's and Gupta's numerical calculations [110] independently led them to a formulation of a criterion analogous to Eqn (11). As shown in Refs [14, 19], expression (11) can be derived as a direct consequence of general criterion (1) by taking the derivative with respect to temperature. If, in addition, τ is considered a function not only of pressure and temperature but also of the corresponding structural order parameters of the system [14, 19], then the memory effect may also be introduced into the model. Note, however, that in our opinion it is expression (1) which is the primary glass transition criterion, and expressions like (11) are merely some of its particular consequences.

3.3 Free-volume models

Underlying the phenomenological free-volume models are the basic hypotheses advanced in Refs [111–114] (see Ref. [3] for the details and description of the early work in this area). The models assume, as is natural, that a certain additional space, a 'free' volume, is always required for the molecular structure to be rearranged. Because the volume occupied by a fluid decreases with cooling, the volume accessible to the molecules decreases, reducing the occurrence probability of the rearrangement necessary for the transition to an equilibrium structure.

Denoting the free volume per molecule by $v_{\rm f}$, the model predicts that the relaxation time can be described via [112]

$$\tau = \tau_0 \exp\left(\frac{C}{v_{\rm f}}\right),\tag{12}$$

where the constant C is material-specific.

The key question in this case is how to define what the free volume accessible to a single molecule in the fluid is. The definitions vary depending on the approaches used. Initially [111], the free volume was defined as the difference between the current volume and the volume the system will occupy after a hypothetical extrapolation to zero temperature; hence,

 $v_{\rm f} \rightarrow 0$ as $T \rightarrow 0$. In another definition, $v_{\rm f}$ is that part of the volume of the fluid which can be rearranged without energy loss. According to Refs [112, 115], $v_{\rm f} \rightarrow 0$ is realized already at a finite temperature, and the relaxation time is given by an expression similar to Eqn (4).

Treating viscosity as simultaneously dependent on the free volume and (directly) on the temperature is discussed in Refs [61, 116–118]. For example, Ref. [116] modifies the initial expression (12) to take into account the activation nature of particle hopping from accessible pockets:

$$\tau = \tau_0 \exp\left(\frac{A}{RT} + \frac{C}{v_{\rm f}}\right) = \tau_0 \exp\left(\frac{A}{RT} + \frac{B}{T - T_{\infty}}\right), \quad (13)$$

where *R* is the molar gas constant, while *A* and *B* are material parameters.

Currently, the free-volume model is rarely applied for modeling purposes because it is now a well-established fact that the relaxation time is not a function of the fluid density alone. Of present-day relevance, though, is the application of its elements to describe the existing data on the structural relaxation of polymers. Thus, Ref. [119] considers modifications of the Adam–Gibbs theory, which are based on the free volume approach, and suggests two expressions for the relaxation time τ , which are used in describing experimental DSC data on polystyrene: the glass transition upon cooling at a rate of 0.1–100 K s⁻¹, and structural relaxation upon holding at 363 K for no longer than 20 h. Compared to the VFT expression, a more accurate fit to the experimental data is achieved.

3.4 Tool-Narayanaswamy-Moynihan method

The Tool–Narayanaswamy–Moynihan (TNM) method based on Tool's concept of fictive temperature and Narayanaswamy's seminal 1971 paper [120] is widely applied even today to analyzing experimental data. Following the publication of Narayanaswamy's paper, Moynihan and colleagues proposed algorithms for calculating the properties of glass under the conditions of isothermal relaxation and temperature being varied at a fixed rate [121, 122] (hence, TNM).

In Ref. [120], the response function of the property p to an instantaneous isobaric temperature change is expressed in the form

$$M_p(t,\Delta T) = \frac{p - p_{2,\infty}}{p_{2,0} - p_{2,\infty}} = \frac{T_{\rm fp} - T_2}{T_2 - T_1} , \qquad (14)$$

where the subscripts 0 and ∞ denote, respectively, the instantaneous and infinite-time values of p; the response function is unity at initial time and zero at infinite time. The nonlinearity of relaxation is reflected in the dependence of M_p on ΔT . The quantity $T_{\rm fp}$ in Eqn (14) is the so-called fictive temperature corresponding to the property p (i.e., different fictive temperatures can be introduced for various properties). The equilibrium response function is defined by the limit of $\Delta T \rightarrow 0$, subject to the departure from equilibrium.

In Ref. [123], Narayanaswamy makes the following three assumptions in his mathematical glass transition kinetics model which necessarily accounts for nonlinearity, nonexponentiality, and memory effects featuring a given process. First, structural relaxation is considered as a single, nonexponential, constant-activation-energy mechanism. Second, the form of the equilibrium response function does not depend on the current value of the system's temperature or on its fictive temperature (an assumption extending the principle of thermorheological simplicity). And third, the driving force and response are related linearly, leaving the dependence of the relaxation time (or viscosity, according to Narayanaswamy) on the structural parameter of the system as the only source of the nonlinearity of the glass transition.

From these three assumptions, one can deduce a linear response function to a finite temperature (T) or structure temperature change. This can be achieved by introducing a reduced time scale ξ :

$$\xi = \int_0^t \varphi(T, T_{\rm fp}) \,\mathrm{d}t'\,,\tag{15}$$

where the function φ describes the time shift of relaxation (or viscosity) according to the empirical formula

$$\ln \varphi(T, T_{\rm fp}) = \ln \frac{\eta}{\eta_{\rm B}} = \frac{H_{\rm g}}{R} \left(\frac{1}{T_{\rm B}} - \frac{1}{T} \right) + \frac{H_{\rm s}}{R} \left(\frac{1}{T_{\rm B}} - \frac{1}{T_{\rm fp}} \right), \quad (16)$$

where $T_{\rm B}$ is a certain reference temperature for the definition of viscosity $\eta_{\rm B}$, while $H_{\rm g}$ and $H_{\rm s}$ are constants equivalent to the activation energies of instantaneous and structural changes, respectively. Using Eqns (15) and (16) and applying the superposition principle (as allowed by the linearity of the response function), the time evolution of the property *p* being considered is given by

$$p(t) - p(0)$$

= $p(0)\alpha_{pl}(T - T_0) - p(0)\alpha_{pe} \int_0^{\xi} M_p(\xi - \xi') \frac{\mathrm{d}T}{\mathrm{d}\xi'} \mathrm{d}\xi', \quad (17)$

where α_{pl} is the change rate of the property p, $\alpha_{pl} \sim (dp/dT)_l$, and α_{pe} is the contribution to α_{pl} owing to the structural relaxation of the system.

In equation (17), the change in p is presented as the difference between its value after complete relaxation and that part of the relaxation which is not achieved over the time passed. The author of Ref. [120] took M_p to be the sum of two or more empirically parameterized exponentials to describe the relaxation of the refractive index in some glasses. The evolution of the heat capacity of the material at the glass transition is described by taking the parameter p to be the enthalpy of the system and usually by assuming that its relaxation time is given by an expression identical to that for the temperature dependence of the viscosity. Currently, the relaxation time is usually written in the modified form [121]

$$\tau(T, T_{\rm f}) = \exp\left[\ln\tau_0 + \frac{\Delta h^*}{RT}x + \frac{\Delta h^*}{RT_{\rm f}}(1-x)\right],\tag{18}$$

where the parameter x is responsible for the nonlinearity of the glass transition, and Δh^* is the material parameter corresponding to the activation enthalpy of the relaxation process.

Narayanaswamy's model [120] was used both by its author [123–127] and by other workers [121, 122]. Moynihan in Refs [122, 128] suggested using Kohlrausch's law as a response function and presented formulas for numerically calculating the evolution of the fictive temperature, irrespective of the temperature history of the system, T(t). For this purpose, the function T(t) is represented as a succession of sufficiently small temperature jumps ΔT_i , which are followed by a pause of duration Δt_i . Then, the fictive temperature at each calculation step is determined by the following formula:

$$T_{\mathrm{f},n} = T_0 + \sum_{i=1}^{n} \Delta T_i \left\{ 1 - \exp\left[-\left(\sum_{j=i}^{n} \frac{\Delta t_j}{\tau_{\mathrm{TNM},j}}\right)^{\beta} \right] \right\}, \quad (19)$$

where $\tau_{\text{TNM},j} = \tau_{\text{TNM}}(T_j, T_{f,j})$.

A series of TNM studies by Hodge and colleagues [57, 58, 129] reported results on the application of the model to DSC experiments on (primarily) polymers, with the thermal history presented either as the aging (quenching) of the sample at a fixed temperature for different time intervals or as uniform cooling at various rates. This work clearly demonstrated the weakness of the model in that a single set of parameters fails to provide a sufficient fit to a set of experimental heat capacity curves. Also, a detailed study of correlations between the model parameters (in particular, x and β) was made, and it was shown, among other things, that in choosing a minimization procedure with respect to a set of parameters, the interrelations between some of the parameters (for example, Δh and x) should be taken into account. Reference [58] also shows correlations between all parameters of the model for five different polymers and two inorganic glasses, which is interpreted as supporting the cooperative nature of the glass transition.

The TNM method described in Ref. [128] can be regarded as a modern formulation of the approach being discussed. The relaxation time τ in expression (19) may be in any form best suitable for the system under study. The proposed method was tested on systems with different sets of properties and for various materials [102, 120, 129] immediately following its proposal.

The TNM method was also applied to model the dynamic glass transition in various systems. References [130-133] considered temperature-modulated calorimetry model experiments on glycerin, polystyrene, polyvinyl acetate, and some mixed systems using the temperature history of the form $T(t) = T_0 + A\cos(\omega t)$ (in early TMC work) or T(t) = $T_0 - qt + A\sin(\omega t)$ (present day TMDSC), respectively, for temperature oscillations around a fixed value T_0 or for linear heating accompanied by temperature oscillations. At sufficiently high temperatures T, the fictive temperature $T_{\rm f}$ has the same profile, but as the average temperature of the system is lowered, the oscillations show a delay in phase and become smaller in amplitude. As a result, the system ceases to respond to ambient temperature oscillations. Characteristically, at sufficiently high frequencies ω this occurs at temperatures T_0 exceeding the glass transition temperature for a given cooling rate. The TNM simulation of these phenomena yields a sufficiently accurate description of some experimental results [130, 131] and makes it possible to consider the analytical frequency dependences of the oscillation amplitude and the deviations in the oscillation phase of polymer characterization parameters [132].

TNM method remains even today one of the most widely used tools for processing experimental data. It should be noted that a similar glass property calculation method was simultaneously developed at the I V Grebenshchikov Institute of Silicate Chemistry of the Russian Academy of Sciences [85, 134, 135]. The reader is referred to the excellent book by O V Mazurin [2] for obtaining further insight into the features of this method and for descriptions of Narayanaswamy's method and the similar Kovacs method [117, 136].

50

3.5 Kovacs method

The method developed in Refs [136, 137] by Kovacs, Aklonis, Hutchinson, and Ramos and known by the abbreviation KAHR can be regarded as an alternative to TNM. Unlike Narayanaswamy's theory, which considers relaxation of only one structural parameter, the fictive temperature T_f , the KAHR model is multiparameteric. As its authors argue, this approach is most adequate for describing the glass transitionrelated memory effect. It should be noted that the four authors also considered simple single-structural-parameter models [117] before they came up with the KAHR model.

There are three basic principles which lie at the heart of the theoretical KAHR method. First, the thermodynamic description of the state of a system is based on the use of order parameters, so different relaxation mechanisms are considered, which operate to bring the system to equilibrium and which are described by a distribution of relaxation times τ_i . This spectrum is assumed to be invariant, meaning that temperature and structural changes do nothing more than shift it along the logarithmic time scale without changing its shape (i.e., similarly to the TNM method, a thermorheologically simple system is considered) [136]. Second, τ_i is allowed to depend both on the temperature and on the instantaneous state (structure) of the glass, yielding the response nonlinear and asymmetric with respect to the degree and sign of departure from equilibrium. And third, the response function is linearized by modifying the time scale in a certain way, i.e., by introducing a new system's time scale determined by the intensity of spontaneous fluctuations in the corresponding equilibrium state.

Thus, the state of a system is determined by a set of parameters (T, P, ζ_i) , $1 \le i \le N$ (here, *P* is pressure), with $\overline{\zeta} = \zeta_i(T, P)$ in the equilibrium state. The system's extensive property *v* (volume, enthalpy) has a total differential of the form

$$dv = \frac{dv}{dT} \left| \frac{dT}{P,\zeta_1,\ldots,\zeta_N} \frac{dv}{dP} \right|_{T,\zeta_1,\ldots,\zeta_N} dP + \frac{dv}{d\zeta_1} \left| \frac{d\zeta_1 + \frac{dv}{d\zeta_N}}{T,P,\zeta_2,\ldots,\zeta_N} \right|_{T,P,\zeta_1,\ldots,\zeta_{N-1}}$$
(20)

In a similar way, we consider the differential for a given property of the system in equilibrium, v_{∞} , subtract the expressions, and introduce a relative departure from equilibrium $\delta = (v - v_{\infty})/v_{\infty}$. The quantity δ can, as conjectured in Ref. [136], be decomposed into separate contributions corresponding to the structural parameters of the system:

$$\sum_{i=1}^{N} \delta_{i} = \delta , \qquad \sum_{i=1}^{N} \mathrm{d}\delta_{i} = \mathrm{d}\delta , \qquad \left(\frac{\partial\delta}{\partial\zeta_{i}}\right)_{T,\zeta_{i}} = \left(\frac{\partial\delta_{i}}{\partial\zeta_{i}}\right)_{T,\zeta_{i}}$$

As a result, the initial equation breaks down after some additional transforms into a system of N kinetic equations for separate departures δ_i . For example, the enthalpy of the system can be written as

$$\frac{\mathrm{d}\delta_i}{\mathrm{d}t} = \Delta C_p q(t) + \frac{\delta_i}{\tau_i}, \qquad \delta = \frac{H - H_\infty}{H_\infty}, \tag{21}$$

where ΔC_p is the heat capacity difference between the glass and the fluid, q(t) is the time dependence of the heating/ cooling rate of the system, H and H_{∞} are, respectively, the system's current and equilibrium enthalpies, and τ_i is the relaxation time spectrum of the system. In Ref. [136], analogous to the TNM method, the response to a jumpwise or linear or arbitrary change in temperature was studied, and the realization of the memory effect in the framework of the method was demonstrated. The behavior of the model $C_p(T)$ curves as functions of experimental parameters and the nonlinearity parameter was given a detailed analysis in Ref. [138].

The initial formulation of the KAHR method [136] assumed that each parameter τ_i depends not on the individual deviations δ_i (or a set $\{\delta_i\}$) but only on the integrated deviation δ :

 $\tau_i = \tau_i(T, \delta) \,. \tag{22}$

Considering the differences between uniform and nonuniform models of the structural relaxation of glasses, the authors of Refs [139–141] discuss, in particular, the applicability of formula (22) to describing relaxation in systems at high degrees of departure from equilibrium. It is shown that, in the framework of assumption (22), the KAHR model is fully equivalent to TNM in terms of the results produced and predictions [141]. The strongly nonlinear kinetics of enthalpy relaxation should expectedly be described by generalizing the dependence in formula (22), i.e., by replacing there the integrated deviation δ by a set of individual deviations { δ_i }. An example of applying the Kovacs method to the glass transition in thin films is given in Ref. [142].

References [143] and [144] carried out KAHR simulations of the dynamic glass transition for, respectively, a system with a single relaxation time and a distribution of τ_i 's; demonstrated the efficiency of the method for describing TMDSC data, and examined the dependence of the components of the complex dielectric constant on the cooling/heating rate, the amplitude and period of temperature oscillations, and on the parameters of nonexponentiality (β) and nonlinearity (x).

3.6 Adam–Gibbs theory

An expression relating the relaxation time to the entropy of the system was obtained in theoretical work by Adam, Gibbs, and DiMarzio [145, 146]. The authors of Ref. [146] presented a statistical theory for the amorphous state of linear molecular chains and, based on their proposed expression for the partition function of a model chain, obtained formulas for the thermodynamic properties of a model polymer. It was shown that, as the temperature is decreased to a certain value T_2 , the model predicts a second-order phase transition to a state characterized by the uniquely accessible chain configuration. While this state is not accessible experimentally, other states (for $T > T_2$, with a larger set of accessible configurations) correspond to glass. Under these assumptions, expressions for the glass transition temperature and for its dependence on the molecular weight, composition, and other properties of the polymer were obtained. Reference [146] later came under criticism: it was noted that the model concept of a chain of molecules relies on some poorly grounded admissions [147], and that there are doubts as to whether the glass transition, a phenomenon common both to polymers and low-molecular fluids, can be explained in terms of the properties of linear chains. Reference [145] subsequently considered extending these ideas (irrespective of polymer structural features) and derived an expression for the relaxation time of the system, which later made it one of the most cited papers in the field.

A transition to an 'ideal' glass state is characterized, as implied by Ref. [146], by the vanishing of the configurational entropy at temperature T_2 . It can, therefore, be assumed that the retarded relaxation of the system in the glass transition region is due to a decrease in the system's configurational entropy (i.e., in the number of accessible polymer configurations) as the transition temperature is approached. The theory is based on the idea of cooperatively rearranging regions of the material, which implies that the system is divided into subsystems weakly interacting with the macroscopic environment. But then it can be considered that over a sufficiently long period of time the system behaves as an ensemble of N independent equivalent subsystems, each consisting of zmolecules. These subsystems are divided into two classes depending on whether cooperative rearrangement is allowed or not. The probability of rearrangement scales as $\sim \exp\left[-z\Delta\mu/(k_{\rm B}T)\right]$, where $\Delta\mu$ is the energy barrier to the rearrangement process (for a single molecule). Clearly, the parameter z equals unity at sufficiently high temperatures, and should approach the entire size of the system as the transition temperature T_2 is approached. The relaxation time τ of the system is inversely proportional to the mean probability of the subsystem rearrangement:

$$\tau = \tau_0 \exp\left(\frac{\Delta\mu S_c^*}{k_B T S_c}\right),\tag{23}$$

where S_c^* is the configurational entropy of the smallest rearrangible ensemble of particles, and the configurational entropy of the system, S_c , can be estimated experimentally as

$$S_{\rm c}(T) = \int_{T_2}^T \frac{\Delta C_p(T)}{T} \,\mathrm{d}T,\tag{24}$$

where $\Delta C_p(T) = C_p^{\text{liq}}(T) - C_p^{\text{gl}}(T)$. The shape of the temperature dependence of ΔC_p determines the expression for the relaxation time; it can be shown, for example, that the hyperbolic behavior of $\Delta C_p(T)$ results in formula (3).

Scherer [148] proposed including the nonlinearity of the glass transition in the sense that the configurational entropy should be dependent on the structure of the system as determined by the fictive temperature at the current instant of time t (rather than by T):

$$S_{\rm c}(T_{\rm f}) = \int_{T_2}^{T_{\rm f}} \frac{\Delta C_p(T)}{T} \, \mathrm{d}T.$$
 (25)

Expressions that follow from the Adam–Gibbs (AG) model can be applied in the TNM framework, i.e., the evolution of the fictive temperature can be traced by substituting various expressions for τ into formula (19). Some authors [148] note that the theoretically grounded AG expression for τ is advantageous over Narayanaswamy's phenomenological expression. Example applications of the method to enthalpy relaxation in polymers are given in Refs [148–152]. Expressions (23) and (24) can also be used with the Kovacs method [117]. Also of note is the fact that the AG model serves as the basis for a number of modern theories, as exemplified by the often-used hypothesis of cooperatively rearranging regions [152–155].

Some alternative to the AG theory consists in developing approaches that use the configurational enthalpy as the structural parameter of the system. By now, a number of phenomenologically augmented models have been proposed based on these approaches, which are outlined in Section 3.7.

3.7 Configurational entropy approach to glass transition

3.7.1 General remarks. The question of whether or not Planck's formulation of the third law of thermodynamics holds for glass was one of the most discussed topics in the science of the glass transition in the 1920s–1930s (as reviewed, for example, in the books [1, 3, 4]). This discussion and the theoretical analysis of the subsequent experimental work led to the main conclusion that at the glass transition the configurational entropy of the system becomes 'frozen-in' at a certain level and then remains unchanged as the temperature of the body tends to absolute zero. The value of T_g —and, hence, the configurational entropy which remained as it was at this T_g —depends on the cooling rate.

Once again under active discussion in recent decades, these well-established results have been, to some extent, thrown into doubt, which renewed intensive discussions on the subject, as exemplified by the International Workshop on Glass and Entropy, Trencin, Slovak Republic (for the proceedings, see the Journal of Non-Crystalline Solids, 355 (2008)). No full consensus exists on the traditional view about the existence of finite entropy at low temperatures. The alternative arguments that attempt to justify a vanishing configurational entropy at the glass transition involve the concept of the continuous violation of the system' ergodicity. As is known, however, the ergodic theory does not supply us with a foundation of the basic assumptions of statistical physics even for equilibrium systems, so its applicability to nonequilibrium systems is even more doubtful. Besides, as is shown in this review (see, for example, the model described in Section 3.8) and elsewhere, the validity of the thermodynamic approach when combined with appropriate statistical models is supported by abundant experimental evidence for the nonzero value of the configurational entropy of glass.

3.7.2 Model involving the limiting configurational entropy S_c^{lim} . The glass transition process and the thermodynamic properties of glass require constructing the model of a glass-forming melt for their theoretical description. A simple example of such a model is considered in Refs [1, 3, 4], which also review more complex approaches to constructing such models. In current section, such models are presented and discussed.

The glass transition model developed in Refs [59, 66, 156– 161] hypothesizes the existence of a limiting 'plateau' in the relaxation of the system for $T < T_g$, an assumption based on the DSC studies of relaxation in various materials at fixed temperatures [162, 163]. The existence of the plateau and the possible two-stage relaxation have been the topics of active discussion over the past twenty years [63–65]. In the AG theory, it is assumed that a certain intermediate metastable state exists in which the system has only a small number of accessible configurations and from which the system cannot move to a lower-energy state within the observationally accessible time scale. The evolution of the configurational entropy in this case is given by

$$S_{\rm c}(t) = S_{\rm c}^{\rm lim}(T(t)) - \sum_{i=1}^{n} \left(S_{\rm c}^{\rm lim}(T_i) - S_{\rm c}^{\rm lim}(T_{i-1}) \right) M(\xi - \xi_{i-1}),$$
(26)

where *M* is the Kohlrausch response function, ξ is the reduced time, the relaxation time is given by the Adam–Gibbs formula (23), and $S_c^{\lim}(T)$ is the maximum accessible value of the



Figure 4. (a) Schematic of the temperature dependence of the configurational entropy for the liquid state (dashed line), experimental dependence for cooling (solid line), and the hypothetic dependence of the limiting state (dashed–dotted line). (b) Plot of heat capacity $C_p(T)$ in a similar notation [59].

configurational entropy. To make a comparison with experimental data, it is necessary to find the temperature dependence of the measured heat capacity, for which purpose the assumption of the equivalent entropy and enthalpy relaxations is imposed on the system. An expression for the evolution of the configurational enthalpy is obtained, from which it is an easy matter to derive an expression for C_p .

In the above-described model, an additional phenomenological parameter δ , the deviation of the metastable state from the extrapolated equilibrium state, necessarily arises (Fig. 4). More specifically, δ determines the deviation of the heat capacity from its equilibrium value C_p (for a supercooled liquid). Further still, the proposal was also made in paper [164] that the temperature T_{gd} at which the metastable state departs from equilibrium be considered as an empirical parameter. While calculations using this model give improved agreement with experimental data, no full agreement has yet been achieved for the entire range of q.

3.7.3 Hypothesis of the nonlinear configurational entropy. An alternative approach, which also relies on the configurational entropy, was developed in Ref. [165], which discusses a method (proposed in Ref. [148]) of introducing nonlinearity into the AG model and proposes an alternative version which takes into account the fact that the configurational entropy is partially frozen-in when a certain fictive temperature is reached. The expression for the total configurational entropy

then takes the form

$$S_{\rm c}(T,T_{\rm f}') = x_s S_{\rm c}(T) + (1-x_s) S_{\rm c}(T_{\rm f}').$$
⁽²⁷⁾

The authors of Ref. [165] argue that, unlike the TNM-model parameter x, which is often thermal-history-dependent, their proposed parameter x_s stands for a material property. The relations between x, x_s , T_f , and T_2 have been obtained. An efficiency comparison [166] of this and the previous model suggests the far-reaching generality of the approach using the method of the limiting configurational entropy.

3.8 Glass transition in terms of nonequilibrium thermodynamics

Instead of employing the methods outlined in Section 3.7 to describe the kinetics of glass transition, an alternative approach based on nonequilibrium thermodynamics can be applied (De Donder, van Rysselberghe [167], Prigogine and Defay [168]), which, apart from using state characterization parameters for a system in classical equilibrium thermodynamics, also introduces additional structural (or internal) order parameters for describing nonequilibrium states of the system. Glass transition treatments within this framework [1–3, 168–171] have been developed based on the ideas formulated by Bragg and Williams [172] in 1934 and by Mandelstam and Leontovich [173]. Our exposition in this review follows the thermodynamics views of Prigogine and Defay, as presented in Refs [168, 169, 174, 175].

By combining the generality of thermodynamics with the ability to use statistical physics models in determining the fine details of the processes and systems studied, the approach we present below provides, in our view, the most suitable tool for general description glass transition kinetics in various materials. The methods under discussion possess the property of flexibility, often absent or not obvious in phenomenological methods, and avoid problems currently encountered in the microscopic description of glass.

Describing glass transition (or, conversely, glass softening) along the lines of de Donder's approach requires introducing at least one additional internal parameter ξ intended to describe the transition of the system to a nonequilibrium state. Generally, ξ is different from the structural parameter ξ_{eq} in the equilibrium state. The dependence of ξ_{eq} on the thermodynamic parameters of the system is determined by considering the assumed model of the system. For example, following Refs [148, 169, 175, 176], one arrives at the simplified expression for ξ_{eq} in the form

$$\xi_{\rm eq} = \begin{cases} g_0(T - T_0) \,, & T_{\rm m} \ge T > T_0 \,, \\ 0 \,, & T_0 \ge T > 0 \,, \end{cases}$$
(28)

where $T_{\rm m}$ is the melting temperature. This model was developed within the framework of the hole theory of liquids [114], and the structural order parameter ξ here corresponds to the number of holes per mole of liquid. A more accurate and more practical expression for $\xi_{\rm eq}$ follows by considering an additional model expression for the thermodynamic potential and entropy [174], and leads to a temperature dependence of $\xi_{\rm eq}$ in the form of a transcendent equation given by

$$\frac{1-\xi_{\rm eq}}{\ln\xi_{\rm eq}} = -\frac{1}{\chi}\frac{T}{T_{\rm m}}\,,\tag{29}$$

with χ being a material parameter.

As the glass transition proceeds, the parameter ξ is 'frozen-in' at values that correspond to T_g , and which then remain unchanged as the temperature is further decreased. The glass softens as it is heated, and the parameter ξ should resume its initial value $\xi_{eq}(p, T)$.

By analogy with introducing the TNM model assumption of the response being linear in the force, the thermodynamics of irreversible processes consider a linear differential equation for the derivative $d\xi/dt$. The range of variability for ξ in some cases is conveniently written as $0 \le \xi \le 1$, where zero corresponds to the ordered state (crystal), and unity to a fully disordered state. The bottom line is the typical generalform equation for ξ :

$$\frac{\mathrm{d}\xi}{\mathrm{d}t} = -\frac{1}{\tau(p,T,\xi)} (\xi - \xi_{\mathrm{eq}}) , \qquad (30)$$

where the system's relaxation time $\tau(p, T, \xi)$ depends (due to the nonlinear nature of the glass transition) not only on the pressure and temperature but also on ξ [175]. Equation (30) is equivalent to equation (9) in Vol'kenshtein–Ptitsyn's theory but differs from it in including the dependence of τ on ξ . The applicability of such equations to the phenomenological description of relaxation process was demonstrated in the most general way by Mandelstam and Leontovich [173].

Having at one's disposal an appropriate expression for $\tau(p, T, \xi)$, the evolution of all the thermodynamic functions of the system undergoing the glass transition can be investigated by solving equation (30). For example, the configurational heat capacity is defined by the expression

$$\Delta C_p(T) = \frac{\mathrm{d}\Delta H(T,\xi)}{\mathrm{d}\xi} \frac{\mathrm{d}\xi}{\mathrm{d}T}, \qquad (31)$$

where the first factor can be considered a constant or can be calculated by using other approximations in combinations with the assumed model of the system. In order to describe cooling or heating of the system, the independent variable in equation (30) is changed from time t to temperature using the heating/cooling rate expression q = dT/dt. Then, the solution of equation (30) yields the temperature dependence of the structural order parameter in the form given in Fig. 1b. Knowing this dependence allows all required thermodynamic quantities to be calculated using appropriately chosen models. For example, by choosing an adequate model for the dependence of the configurational entropy on the structural parameter, a dependence similar to that shown in Fig. 5a can be obtained [17] for the 'frozen-in' configurational entropy (which remains unchanged as the system is cooled to $T \rightarrow 0$). Evidently, this simple model is consistent with the conclusion derived from a huge amount of experimental data showing that the configurational entropy is frozen-in at $T_{\rm g}$, and that its value for $T < T_g$ is thus determined by the cooling rate.

Shown in Fig. 5b, c are the dependence of the configurational entropy S_c on the structural parameter ξ and the temperature dependence of the equilibrium value S_c^{eq} , both calculated within the hole model of the glass-forming liquid. It is seen that for the appropriately chosen model of the system, no Kauzmann paradox [84] arises and that the configurational entropy of the metastable system tends to zero only when $T \rightarrow 0$. In terms of this model, the paradox will occur only if an approximation inadequate for temperatures $T \ll T_g$ is chosen, such as the linear approximation adopted in Eqn (28) (see also Ref. [2]).



Figure 5. (a) Cooling rate dependence of the 'frozen-in' configurational entropy of a model system. The dependences of the configurational part of the entropy on the structural parameter ξ (b) and on the reduced temperature (c) calculated from the model expressions for S_c and ξ_{eq} [17].

Another interesting results consist in the estimate of the entropy produced in the system during its irreversible relaxation. The entropy production over the glass transition region is nonzero, similar to the case with the imaginary part of the response function at the dynamic glass transition (see Eqn (5) and Fig. 2). The results shown in Fig. 6 [17, 174] are obtained using the equation

$$\frac{\mathbf{d}_i S}{\mathbf{d}T} = \frac{G_{\mathrm{eq}}^{(2)}}{q T \tau} \left(\xi - \xi_{\mathrm{eq}} \right)^2, \qquad G_{\mathrm{eq}}^{(2)} = \frac{\partial^2 G}{\partial \xi^2} \Big|_{\xi = \xi_{\mathrm{eq}}}, \tag{32}$$



Figure 6. Entropy production in a system undergoing the glass transition and softening. $\theta = T/T_m$ is the reduced temperature. Lines *I* and *2* are for cooling and heating of the system, respectively [17].

where G is the Gibbs energy of the system. The square of the difference $\xi - \xi_{eq}$ in Eqn (32) ensures that the entropy of the system increases irrespective of how the irreversible process proceeds. The plot in Fig. 6 of the temperature dependence of the entropy production d_iS in the course of the heating and cooling processes clearly shows the width of the glass transition region in thermodynamic terms. Thus, it is shown in Ref. [177] that this width is determined by the amount of entropy produced at temperature T_g during cooling. See below for more on this.

The results presented above rely on the VFT relaxation time expression (4). Fitting the simulation results to the experimental data may require various modifications of the expression for τ in equation (30). In particular, it was shown in paper [175] that an equation of the form

$$\frac{d\xi}{dt} = -\frac{\xi - \xi_{eq}}{\tau_{eq}(T)} \exp\left[-\frac{b_2}{RT}(\xi - \xi_{eq})\right]$$
(33)

is equivalent to the TNM model with the function of form (16) proposed by Narayanaswamy. A nonexponential form for the relaxation of the system in the Kohlrausch form is obtained by simply writing equation (30) as follows:

$$\frac{\mathrm{d}\xi}{\mathrm{d}t} = -\frac{1}{\tau_{\mathrm{eq}}(p,T)}\,\varphi(t)(\xi - \xi_{\mathrm{eq}})\,,\tag{34}$$

where $\varphi(t) = A_0^*/t^{1-\beta}$, with A_0^* being a constant [175].

The approach we have presented can be extended to cases where several structural parameters are needed to describe the state of a glass-forming system. The evolution of the system can then be described (in the linear approximation) by the system of kinetic equations in the form

$$\frac{d\xi_j}{dt} = -\sum_{i=1}^f \frac{1}{\tau_{ij}(p,T)} (\xi_i - \xi_i^{eq}), \qquad (35)$$

which, by introducing the effective relaxation time of the *j*th structural parameter ξ_j , can be written out as

$$\frac{\mathrm{d}\xi_j}{\mathrm{d}t} = -\frac{1}{\tau_j^{\mathrm{eff}}(p, T, \{\xi_i\})} (\xi_j - \xi_j^{\mathrm{eq}}), \qquad (36)$$

where

$$\frac{1}{\tau_j^{\text{eff}}(p, T, \{\xi_i\})} = \sum_{i=1}^f \frac{1}{\tau_{ij}(p, T)} \frac{\xi_i - \xi_i^{\text{eq}}}{\xi_j - \xi_j^{\text{eq}}}.$$
(37)

The new parameter $\tau_j^{\text{eff}}(p, T, \{\xi_i\})$ should then be substituted into the corresponding expressions, for example, the one for determining the glass transition temperature [14].

Experimentally, the glass transition process can be measured and investigated by a jumpwise change in the thermodynamic coefficients of the system, such as the heat capacity, heat expansion coefficient, and compressibility at T_g . By combining them, the well-known Prigogine–Defay ratio (an analog of the Ehrenfest relations) at the glass transition can be obtained in the following form [1, 3, 4]

$$\Pi(T_{\rm g}) = \frac{1}{VT} \frac{\Delta C_p \Delta \kappa_T}{\left(\Delta \alpha_p\right)^2} \Big|_{T=T_{\rm g}},\tag{38}$$

where ΔC_p is the difference between the isobaric heat capacities of the glass-forming liquid and that of glass, while $\Delta \kappa_T$ and $\Delta \alpha_p$ are the respective differences in the isothermal compressibilities and thermal expansion coefficients. For measurements on glasses, typically, $\Pi > 1$, which differs from the Ehrenfest relations for second-order phase transitions in which the analogous parameter equals unity. The problems of how, in principle, to derive this parameter from Eqn (38) and how to obtain its values for different models are the subject of much theoretical discussion [104, 174, 178, 179]. A simplistic view of the glass transition (Simon's model) suggests that the Prigogine–Defay ratio is always unity, provided the description of the transition utilizes only one structural order parameter and the glass transition itself is considered to be instantaneous at $T = T_g$. However, as shown, for example, in Refs [4, 179], if one takes into consideration that the glass transition occurs over a range of temperatures, then the expression for ratio (38) can differ from unity even if the system is modelled by a single structural parameter ξ . An exact expression for Π was derived in paper [18], which can be well-approximated by a relation first formulated in Ref. [174] and having the form

$$\Pi(T_{\rm g}) \approx \frac{h_{p,T}}{A + h_{p,T}} \bigg|_{T = T_{\rm g}},\tag{39}$$

where A is the affinity corresponding to the parameter ξ , and $h_{p,T}$ is the corresponding derivative of the enthalpy, $h_{p,T} = (\partial H/\partial \xi)_{p,T}$. The value of A in Eqn (39) should be calculated for cooling conditions, resulting in A < 0 and $\Pi > 1$. Values of $\Pi > 1$ can also be obtained by directly considering the thermodynamic properties of equilibrium liquid and glass. See paper [14] for various approaches to this problem. What makes these results fundamentally important is that a single structural parameter is sufficient for describing glass-forming liquids. This does not mean, of course, that this is always so, but it can be concluded that single-parameter models are quite acceptable for describing glass-forming systems.

3.9 Some current microscopic approaches to glass transition description

The development of microscopic theoretical approaches to describing the glass transition is highly topical and an actively growing area in modern glass science. Describing in statistical physics terms the transition of a liquid into an amorphous vitreous state is a problem of ever-increasing interest, it being hoped that future theories will be advantageous in a number of respects — in particular, that in many cases no empirical parameters will be needed to make predictions. Applying the apparatus of statistical physics allows a detailed comparison to be made both with scattering experiments and with computer simulations. The reverse is also possible: experimental data (on the structural factor, for example) or the results of detailed computer simulations can be used as the basis for a subsequent, sufficiently complete simulation of various properties of a system undergoing the glass transition.

At the same time, it should be noted that the statistical theory of supercooled liquids and that of glass formation process are not yet complete, which is manifested by the fact that a correct quantitative description is not available and phenomenological corrections should be introduced into the models. Another point is that classifying the theories we present below as modern is rather vague: by modern, we mean here those which, as believed, are amenable to a great future development. The approaches described in Sections 3.1–3.8 are worked out to their full extent and cannot be improved other than by considering corrections or phenomenological expressions for some of the system's properties. Notice also that the 'microscopic' methods presented below can only be ambiguously termed so because of the mean field approximation they often use.

Sections 3.9.1–3.9.3 briefly summarize some basic ideas behind current microscopic theories of the glass transition. Detailed information can be found in specialized reviews cited in the text, where needed. Also reviewed are a number of studies which, among other direct results, simulate the heat capacity curves of a system undergoing the glass transition.

3.9.1 Energy landscape model. The employment of the system's energy landscape as a basis for simulating the glass transition and glass properties was pioneered by Goldstein [180], who was the first to indicate the importance of the energy surface topography for the dynamics of glass-forming material at low temperatures (note that the discussion of phase space structure during the glass transition dates back to Kauzmann [84]). Simulation in this case evolves on the base of qualitative—in time, hopefully, quantitative—description of the multiparticle function $\Phi(\mathbf{R})$, the potential energy of a system with a configuration as specified by the full set of position vectors $\{\mathbf{R}\}$ of its constituent particles. The behavior of a system as, for example, it is cooled down from the melting point resembles, in dynamics terms, the motion of a particle over the hypersurface Φ . Correspondingly, this expression is determined by the topography of the hypersurface $\Phi(\mathbf{R})$ which, in turn, is determined by the properties of the system (for example, by the chemical nature of its constituents). Topography can be thought of as a set of such elements as local and global minima and saddle points of various orders [181, 182]. The key concept here is the so-called 'basin' [90, 181, 182], a region in configuration space which surrounds one of the potential minima and brings the system to the state corresponding to this minimum. Clearly, a global minimum on this surface corresponds to the crystalline state; moreover, there are a large number of minima that are 'close' to it and which correspond to crystals with defects and dislocations or to polycrystals.

In addition to the minima mentioned there are, however, many other states — and these will be discussed below — that

correspond to a system structure with no long-range order, i.e., a system in an amorphous state (glass). The number of these minima increases exponentially with the number of particles N in the system. If the liquid is supercooled, then, as some of the minima become inaccessible, the function $S_{\rm c} = k_{\rm B} \ln N_{\rm c}(T)$ decreases ($N_{\rm c}$ is the number of accessible basins). Further, the state of the system is determined by the statistical depth distribution of the minima, so that the temperature of the system determines the probability of finding the system in the basin of a given depth. In this framework, the differences between 'fragile' and 'strong' glass-forming materials can be expressed in terms of the differences in the topography of the surface $\Phi(\mathbf{R})$ [90]. It can also be shown [183] that increasing the cooling rate results in the system being 'locked' in the higher-energy basin, followed by its subsequent transition to lower-energy states, i.e., minima of the surface $\Phi(\mathbf{R})$ (structural relaxation). In this case, the dynamics of the transition are also determined by the topography of $\Phi(\mathbf{R})$.

One approach [153, 184] to simulating the glass transition of the system consists in modeling the landscape of the free energy $\Phi(T, V, N, \{\mathbf{R}_i\})$. The probability of finding the system in a certain configuration $\{\mathbf{R}_i\}$ is then given by

$$P(\{\mathbf{R}_i\}) = \frac{\exp\left[-\beta\Phi(T, V, N, \{\mathbf{R}_i\})\right]}{Z(T, V, N)}, \qquad (40)$$

where Z is the partition function of the system. The expectation value of a certain physical quantity A is, by definition, the integral of the product AP over all configurations $\{\mathbf{R}_i\}$. Hence, the derivative of A with respect to temperature is partly determined by the topography of the surface Φ . For example, Ref. [184] examines the entropy and heat capacity of the system:

$$C = \left\langle T \frac{\partial S(\{\mathbf{R}_i\})}{\partial T} \right\rangle + \frac{1}{k_{\rm B}T^2} \left\langle \Delta E(\{\mathbf{R}_i\})^2 \right\rangle, \tag{41}$$

where $\Delta E({\mathbf{R}_i}) = E({\mathbf{R}_i}) - \langle E \rangle$, and *E* is the energy of the system in the configuration $\{\mathbf{R}_i\}$, and uses a simple model as an example to demonstrate the significance of the second term in Eqn (41) in the description of experiments. The study of the dynamic properties of the system requires introducing an additional probability function to take into account the initial and final configurations of the system. The dynamics of Φ for a simple model is considered in Ref. [185], whereas another approach is studied in Refs [186-189]. Calculations of the evolution of heat capacity curves were performed for model systems in Refs [190-194]. Reference [194], in particular, reported the dependence of the glass transition temperature on the cooling rate. References [62, 195] calculate the evolution of the heat capacity of polystyrene and polymethyl methacrylate (PMMA) with a model corrected by introducing a temperature-dependent Φ . Reference [196] discusses the general aspects of the temperature dependence of the relaxation time by applying the energy landscape models to polymers.

A series of studies [197–199] present a model allowing one to calculate the enthalpy hypersurface for describing the glass transition in selenium. At high temperatures, the system is easy to move between various basins of the hypersurface H, but upon cooling such transitions between different structures (basins) become rarer until finally the system in the glassy state turns out to be 'locked' in a phase subspace. The probability for the system to assume a given configuration (to occupy a given basin) is determined by an expression analogous to formula (40). The dynamics of the enthalpy landscape is described by a set of differential equations (whose number is equal to the number of basins in H), their coefficients being determined by the thermal history and the theory of transition states. In Refs [197–199], an exact calculation of the function H for a system of a small number of particles was performed using *ab initio*-calculated interatomic potentials for selenium, providing the description of the evolution of the molar volume, thermal expansion coefficient, and viscosity of a system under cooling.

3.9.2 Mode-coupling theory. One of the popular theories that have been developed in recent decades to describe the dynamics of supercooled liquids is the mode-coupling theory (MCT), first proposed independently in Refs [200] and [201]. At the basis of MCT lies the equation of motion for the dynamic structure factor F(k, t) of a one-component liquid:

$$\frac{d^2 F(k,t)}{dt^2} + \frac{k^2 k_{\rm B} T}{mS(k)} F(k,t) + \int_0^t d\tau \, M(k,\tau) \, \frac{d}{dt} F(k,t-\tau) = 0 \,.$$
(42)

The exact description of the dynamics of the system requires the specification in equation (42) of the quantities $M(k, \tau)$, the memory kernel for particles of a given mass at a given temperature, and S(k) = F(k, 0), the static structure factor. The fundamental assumption of MCT is the closure in the expression for $M(k, \tau)$ to the autocorrelation density functions F(k, t) written in a quadratic form [201]. Thus, knowing the specific dependence S(k) for the liquid under consideration, numerical methods can be applied to solve equation (42). An important prediction of MCT is the value of the critical temperature T_c at which the self-diffusion coefficient becomes zero or, in other words, an infinite increase is observed in the α -relaxation time of the correlation functions:

$$D \sim \frac{1}{\tau} \sim (T - T_{\rm c})^{\gamma}, \qquad (43)$$

where the exponent $\gamma > 3/2$. This prediction has been tested repeatedly in many experiments and computer simulations, and it has been shown that in some systems *D* and τ follow different power laws. It is also shown in paper [202] that *D* cannot become zero (this dependence is an artifact of the idealized MCT).

Of interest is the character of the time dependence of the reduced correlation function $\Phi(t) = F(k_0, t)/S(k_0)$ (see Fig. 7). There are three regimes which can be recognized



Figure 7. Time evolution of the correlation functions $\Phi(t)$ according to MCT for different temperatures [204, 206].

for different temperatures: the initial fast relaxation to a plateau which is determined by the value of k_0 ; the slow relaxation near the plateau, and the β -relaxation and structural α -relaxation.

The explanation for this behavior of $\Phi(t)$ is that over a short period of time the particle moves ballistically, but then the particle turns out to be 'locked' in a 'cage' of its nearest neighbors. It is not until some time later, when the β -relaxation is completed, that the particle leaves the cage. Yet another prediction of MCT is the superposition principle for α -relaxation:

$$\Phi(t) = \Psi\left(\frac{t}{\tau(T)}\right),\tag{44}$$

i.e., correlations for different temperatures can be reduced to a single curve by plotting them on the t/τ scale. Also of interest is the fact that the MCT prediction for Ψ in Eqn (44) corresponds to the Kohlrausch formula.

A characteristic feature of the so-called idealized MCT model, the low temperature behavior of correlation functions plotted in Fig. 7, does not fit the reality. An extended version of MCT considers the possibility of activation processes accompanied by a particle hopping between the 'cages'. Thus, Ref. [203] shows that such an approach restores the ergodicity of the system at and below T_c . Some details of MCT are reviewed in Refs [87, 204–207].

Overall, despite some limitations and disadvantages, MCT is an effective analyzing tool for glass-forming materials, providing general insights based on microscopic interaction potentials between the atoms and molecules of the system under study [208].

3.9.3 RFOT theory. The work of Kirkpatrick, Thirumalai, Wolynes, and others over the last few decades has resulted in a new microscopic approach, which its authors called the Random First-Order Transition (RFOT) theory [209–211] and which is, in fact, the application of a similar previously developed theory of transitions in model spin systems [11, 212] to the glass transition [213, 214]. In the development of their approach, the authors also heavily relied on the Adam–Gibbs theory, MCT, and the energy landscape model.

The term 'random first-order transition' is meant to refer to the freezing of a liquid and its conversion to a number of aperiodic structures (quasicrystals), a point which sets this transition fundamentally apart from first- and second-order phase transitions. The authors of Refs [209–211] note the existence of two temperatures which determine the glass transition behavior of the system. For a temperature Tbelow the dynamic transition temperature T_A , it can be shown in the mean field approximation that part of the phase space of the system is a set of a large number of potential 'wells' corresponding to the statistically equivalent amorphous states (similar to Goldstein's theory, the number of such states increases exponentially, ~ exp (αN), where N is the number of particles in the given region of the system).

For $T < T_A$, the displacements of molecules in a liquid are primarily determined by activation processes, leading to the 'stratification' of the system and to its transformation into a mosaic-like system of 'drops' in various amorphous states (as in the AG theory, the size of these regions is characterized by the correlation length ξ). The authors of the RFOT theory followed the analogy with the classical theory of nucleation [215–217] in taking into account the difference in the



Figure 8. (Color online.) RFOT/MCT simulation of temperature curves C_p^{red} [see Eqn (8) in the text] for glycerin for different cooling rates q marked in panels [223].

rearrangement due to the phase interface in the system (which is where this theory differs from the AG theory). However, whereas in the nucleation process (which leads to a first-order phase transition) the formation of a sufficiently large number of particles in the new phase causes a fast growth of clusters at a given temperature, the tessellation model of a cooled liquid allows the formation of a new 'drop' of the amorphous phase within the already existing larger 'drop'. This leads to the formation of a system of clusters that are in the process of continuous rearrangement and whose size increases only with decreasing temperature according to the law

$$\xi \sim \frac{1}{\left(T - T_{\rm K}\right)^{2/d}} \,. \tag{45}$$

The temperature $T_{\rm K}$ in formula (45) is the temperature of a random first-order phase transition (to the state of 'ideal' glass), a temperature at which the configurational entropy determining the transition between amorphous states becomes zero. The divergence in formula (45) is stronger than in the Adam-Gibbs theory, a fact which in some systems should result in the amorphous clusters becoming nanometer sized. The RFOT theory formulated in this way, as well as the AG model, is currently associated in a certain sense with the discovery in glass-forming systems of nontrivial spatiotemporal fluctuations of a comparable size, called dynamic heterogeneities [206, 218, 219]. Over the last twenty years, much experimental, theoretical, and computer simulation work has been done to study these heterogeneities [206, 218, 220]. Some studies, though, have cast doubt on the existence of this effect in molecular liquids [221, 222].

An example of the application of the RFOT theory (in combination with MCT) to glass formation kinetics is given in Ref. [223], which yields the expression for the mobility field

 $\mu(x, t)$ of a typical form employed in the system of equations for the fictive temperature:

$$\frac{\partial T_{\rm f}}{\partial t} = -\mu(T_{\rm f} - T)$$

Reference [223] simulates the heat capacity curves for glycerin and three other glass-forming materials for different heating and cooling rates (Fig. 8). This work also shows that the theory under review produces a bimodal distribution of relaxation times for a polymer (polystyrene) at structural relaxation at temperatures well below T_{g} .

Overall, we note that, although the RFOT theory, which is a conglomerate of various theories, is advantageous in making many analytical predictions, it remains disadvantageous in some respects and also leads to new problems, such as the description of the transition from the dynamical slowing-down in the MCT regime to the state involving the proceeding of activation processes.

4. Kinetic criteria of glass transitions

4.1 General kinetic criterion of glass transition and examples of its application

Establishing the dependence of glass properties on the heating or cooling rate is important for practical applications. Furthermore, the very fact of the existence of this dependence is fundamentally important for the correct description of the glassy state and the glass formation process. The Bartenev–Ritland law, Eqn (3), is the first result in this area [20, 22]. In a subsequent paper [176], a similar analysis by Moynihan confirmed the experimental validity of this law for three different glasses and also stated that the law applies in general to glass-forming systems. A similar statement was recently made by Mazurin [224]. A detailed analysis of the problem was carried out in monograph [1], and Refs [17, 18, 225] present numerical simulations for a wide range of cooling and heating rates.

In paper [19], a general kinetic criterion of glass transition is formulated for the case of a temperature variation, as well as of other external control parameters generalizing, in particular, earlier work in this direction. The glass transition, i.e., the kinetic transition of a system from equilibrium state to nonequilibrium, can be described in a general way in the framework of the thermodynamics of nonequilibrium processes by adding a set of state parameters $\{\xi_i\}$ to the usual properties of the system. When the system requires no more than one structural parameter ξ to describe its nonequilibrium state, the kinetics of the change in this parameter under isothermal conditions is described in a general way by an equation of the form (30). If the system is heated or cooled at a fixed rate q, then one can write the following equation

$$\frac{\mathrm{d}\xi}{\mathrm{d}T} = -\frac{1}{q\tau(p,T,\xi)} \left(\xi - \xi_{\mathrm{eq}}\right), \qquad q = \frac{\mathrm{d}T}{\mathrm{d}t}.$$
(46)

By introducing the characteristic time for variation in temperature τ_T , the change in the temperature of the system can be written in a form similar to Eqn (30):

$$\frac{\mathrm{d}T}{\mathrm{d}t} = -\frac{1}{\tau_T} T, \qquad \tau_T = \frac{T}{|q|} \,. \tag{47}$$

We can use the relation between times τ and τ_T to consider the stages a liquid goes through in the process of glass transition under cooling (or in the reverse process). Clearly, the system can remain in a state of thermodynamic equilibrium if $\tau \ll \tau_T$, whereas, in the opposite extreme, $\tau \gg \tau_T$, the state of the system is 'frozen-in', meaning that the glass transition is completed. Hence, the glass transition interval is determined by the relation $\tau \sim \tau_T$, i.e., the characteristic times for the relaxation of the system and for temperature change are of the same order of magnitude. Thus, we have arrived at a general formulation of the kinetic glass transition criterion. With this criterion, the expression for the glass transition temperature can be written as [14, 19]

$$\tau \approx \tau_T, \qquad \left(\frac{1}{T} \left|\frac{\mathrm{d}T}{\mathrm{d}t}\right| \tau\right) \Big|_{T=T_{\mathrm{g}}} \approx C, \qquad C \approx 1.$$
 (48)

The selected value of C in Eqn (48) should remain unchanged when determining T_g for various cooling or heating rates. The expression $\tau \sim \tau_T$ can be applied in the general form to the glass transition when other properties of the system, for example, pressure, are varied, or it can be used to determine the temperature of the dynamic glass transition.

Clearly, expression (48) can be approximated by $(|q|\tau_R)|_{T=T_g} = \text{const}$ (according to Bartenev's proposal [20, 21]) or by using Maxwell type relaxation processes in the form of the Bartenev–Ritland law (3). It is also possible, however, to obtain a generalized Bartenev–Ritland law, for example, the relationship

$$\frac{1}{\tilde{T}_{g} - \tilde{T}_{0}} = \ln \tilde{T}_{g}^{2} - \ln \tilde{q}, \qquad (49)$$

if VFT relaxation is assumed. Equation (49) uses the reduced values of the temperature and cooling rates [19, 177]. Expression (49), whose right-hand side is seen to directly depend on $T_{\rm g}$, suggests that for certain values of the cooling

rate q the behavior of T_g deviates from that given by the original expression of the Bartenev–Ritland law in the form of Eqn (2) (see Section 4.2 for details).

From Eqn (48) also follows an expression for the change in the glass transition temperature T_g as a function of other system parameters, for example, pressure [14, 19]:

$$\frac{\mathrm{d}T_{g}}{\mathrm{d}p} = -\frac{(\mathrm{d}\tau/\mathrm{d}p)_{T} + (\partial\tau/\partial\xi)(\partial\xi/\partial p)}{(\mathrm{d}\tau/\mathrm{d}T)_{p} - \tau/T}\Big|_{T=T_{g}, \ p=p_{g}, \ \xi=\xi(T_{g}, p_{g})}.$$
(50)

Hence, determining the derivative dT_g/dp requires knowing the dependences $\tau(p, T, \xi)$ and $\xi(p, T)$. Formula (50) is readily extended to the case of several structural parameters $\{\xi_i\}$ [14]. In the simplest case, various expressions can be obtained for dT_g/dp by neglecting the pressure dependence of the structural order parameters and using different models for $\tau(p, T)$. Like Ehrenfest's theory of second-order phase transitions, here we do not have two equations that are always applicable for dT/dp, but only one general equation, Eqn (50). Hence, the pressure dependence of the glass transition temperature does not allow us to conclude that the Prigogine–Defay ratio is equal to unity [14, 19].

Assuming linearity between the relaxation time and the viscosity of the liquid (which is true for many glass-forming materials), we can derive an expression for determining η at the glass transition temperature [14, 17, 18]:

$$\log \eta(T_{\rm g}) \approx B - \log |q|, \qquad (51)$$

where the parameter *B* is weakly dependent on T_g . It follows that Tammann's [24] popular and widely used glass transition criterion assigning a definite value of viscosity to T_g does not apply to a wide range of heating (cooling) rates used in the process. Given the rate, the value of viscosity at the glass transition can be calculated by Eqn (51), where the temperature T_g is determined from the general criterion (48).

As an example of the application of criterion (48) to the dynamic glass transition, let us consider the relaxation of a system under the condition where the external parameter (temperature) oscillates with frequency ω . The equilibrium value of the structural parameter in Eqn (46) varies as $\xi_{eq} \sim \exp(i\omega t)$. But then, by considering the characteristic time $\tau_D = 1/\omega$ for the change in the equilibrium value ξ_{eq} , we can obtain a typical criterion (denoted frequently as Frenkel–Kobeko relation [3, 4]) of the form $\omega \tau \sim C$, where $C \sim 1$ [19, 20]. In order to arrive at the same value of T_g at thermal and dynamic glass transitions (determined via the Frenkel–Kobeko relation) the condition $q/(T\omega) = C \approx 1$ must be fulfilled [114, 226].

Knowing the evolution of the structural parameter ξ and taking into account the evolution of p and T makes it possible to calculate the evolution of any other property of the system. Assuming that one of the system parameters, φ_m , is nonlinear in ξ , $\varphi_m \propto (\xi - \xi_{eq})^m$, and using the Kohlrausch form for the relaxation of ξ , namely

$$\frac{\xi(t) - \xi_{\rm eq}}{\xi(0) - \xi_{\rm eq}} = \exp\left[-\left(\frac{t}{\tau}\right)^{\beta}\right], \quad \beta < 1,$$
(52)

the relaxation of φ_m is readily expressed as [14]

$$\ln \varphi_m(t) = \ln \varphi_m(0) - \left(\frac{t}{\tau^{(m)}}\right)^{\beta}, \quad \tau^{(m)} = \frac{\tau}{m^{1/\beta}}.$$
 (53)

Hence, it can be concluded that different thermodynamic properties of a system may have different relaxation times, i.e., the calculated values of the glass transition temperature T_g depend not only on the material under study or the rate of change of the external parameter but also on the specific property which is being determined in the experiment. Further, assuming that the relaxation time of the parameter ξ is described by the VFT law (4) and using criterion (48), we can obtain the modified Bartenev law

$$\frac{B}{T_{\rm g} - T_0} = \log \frac{T_{\rm g}}{\tau_0} - \log |q| \,. \tag{54}$$

Similarly, for the property φ_m , making the replacement $\tau_0 \rightarrow \tau_0/m$, we have

$$\frac{B}{T_{\rm g}^{(m)} - T_0} = \log \frac{T_{\rm g}^{(m)}}{\tau_0} + \log m - \log |q| \,.$$
(55)

Equations (54) and (55) express the behavior of the glass transition temperatures obtained by various methods for different cooling rates of the system. It is seen that the value of *m* determines the difference between T_g and $T_g^{(m)}$ for equal values of *q*.

4.2 Width of a glass transition range

The approach described in Section 4.1 can be applied to the general analysis of the glass transition width as a function of the heating or cooling rate of the system [177]. For the temperatures characterizing the boundaries of the glass transition interval, we can write

$$T_{g}^{+}: \quad \left(\frac{1}{T} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right| \tau \right) \Big|_{T=T_{g}^{+}} \approx \alpha^{+} \ll 1 ,$$

$$T_{g}^{-}: \quad \left(\frac{1}{T} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right| \tau \right) \Big|_{T=T_{g}^{-}} \approx \alpha^{-} \gg 1 .$$
(56)

The glass transition width depends on the choice of the parameters α , which fully corresponds to the methodological features of any its experimental or theoretical definition [110]. Using relations (56), one can obtain analytical expressions for the glass transition width for systems whose relaxation is expressed by two widely employed laws, the Arrhenius law and VFT phenomenological equation.

Let us write the Arrhenius law in the form

$$\tau = \tau_0 \exp \frac{U_0}{RT} = \tau_0 \exp \frac{A}{T}, \qquad \tau_0 = \frac{h}{k_{\rm B}T}.$$
(57)

Then, by substituting the expression for $\tau(T)$ into Eqns (48) and (56) and introducing the dimensionless variables for the temperature and temperature change rate, viz.

$$\tilde{T} = \frac{T}{A}, \qquad \tilde{q} = \frac{h}{k_{\rm B}A^2} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right|,$$
(58)

the relation for the glass transition temperature \tilde{T}_{g} can be written as

$$\frac{1}{\tilde{T}_{g}} = \ln \tilde{T}_{g}^{2} - \ln \tilde{q} \,. \tag{59}$$

Noting that the logarithm of the square of the reduced glass transition temperature varies much more slowly than the



Figure 9. (Color online.) Comparison of characteristic reduced temperatures of the glass transition interval as a function of the cooling/heating rate for the Arrhenius (dark lines) and VFT (blue lines) relaxation times [177].

logarithm of q, expression (59) is an analog of Bartenev's glass transition law.

For the glass transition boundaries, we similarly find

$$\frac{1}{\tilde{T}_{g}^{+}} = \ln\left[(\tilde{T}_{g}^{+})^{2}\alpha^{+}\right] - \ln\tilde{q},$$

$$\frac{1}{\tilde{T}_{g}^{-}} = \ln\left[(\tilde{T}_{g}^{-})^{2}\alpha^{-}\right] - \ln\tilde{q}.$$
(60)

Using Eqns (59) and (60) with the specified values of α^{\pm} , it is possible to investigate analytically T_g and the glass transition interval width as a function of the cooling rate q over its entire range. The curves of the glass transition temperature and glass transition interval boundaries are shown in Fig. 9 for a wide range of q.

Introducing similar reduced quantities

$$\tilde{T} = \frac{T}{B}, \qquad \tilde{T}_0 = \frac{T_0}{B}, \qquad \tilde{q} = \frac{h}{k_{\rm B}B^2} \left| \frac{\mathrm{d}T}{\mathrm{d}t} \right|, \tag{61}$$

for the Vogel–Fulcher–Tammann relaxation law (4), the dimensionless glass transition temperature and the upper and lower boundaries of the glass transition interval can be expressed as [177]

$$\frac{1}{\tilde{T}_{g} - \tilde{T}_{0}} = \ln \tilde{T}_{g}^{2} - \ln \tilde{q},$$

$$\frac{1}{\tilde{T}_{g}^{+} - \tilde{T}_{0}} = \ln \left[(\tilde{T}_{g}^{+})^{2} \alpha^{+} \right] - \ln \tilde{q},$$

$$\frac{1}{\tilde{T}_{g}^{-} - \tilde{T}_{0}} = \ln \left[(\tilde{T}_{g}^{-})^{2} \alpha^{-} \right] - \ln \tilde{q}.$$
(62)

The first expression in Eqn (62) is (with the above reservations) the modified Bartenev–Ritland equation for the VFT relaxation law. How the relaxation law affects the glass transition temperature can be assessed by comparing the dependences obtained from Eqns (59), (60), and (62) (see Fig. 9). It is seen that differences already appear at sufficiently low cooling/heating rates, i.e, when the glass transition occurs at temperatures sufficiently close to T_0 . A comparison of analytical calculations and numerical simulations using Eqn (46) [177] shows that, for the dependences to be identical, it is necessary to select such values of α^+ and α^- that are probably determined by the parameters of the relaxation law and by other properties of the simulated system.

The glass transition width can also be estimated in an alternative way to that described above by employing only the general thermodynamic expressions for the properties of a system undergoing a glass transition. We can take the dependence of the entropy production in the system as the basis for this: it is seen from Fig. 6 that outside of the glass transition interval it is zero, allowing this quantity to be used for deriving expressions for ΔT_g . The position of the maximum of d_iS under cooling can then be considered as the thermodynamic definition of the glass transition temperature, and the width of the interval is determined by Taylor expanding d_iS/dT [177]. As a result, describing the relaxation by equation (46), the following estimates may be obtained:

$$T_{g} \approx \left[\frac{d}{dT}\ln\frac{(\xi - \xi_{eq})^{2}}{\xi_{eq}}\right]^{-1},$$

$$T_{g}^{+} - T_{g}^{-} = 2\left\{-\frac{1}{2}\frac{d^{2}}{dT^{2}}\left[\ln\frac{1}{q\tau}\frac{(\xi - \xi_{eq})^{2}}{\xi_{eq}}\right]\Big|_{T=T_{g}}\right\}^{-1/2},$$
(63)

which can be used to directly calculate the glass transition temperature and the glass transition interval width for systems with a single structural parameter (assuming the temperature dependences of ξ , ξ_{eq} , and τ are known).

5. Physical relaxation of polymers

The structural relaxation of glass-forming liquids and polymers represents a kinetic process which is currently of topical interest from both a fundamental perspective (description of the glass transition process) and for a variety of applications. The fact that materials dynamically change their properties (for example, viscosity) when stored or used at a given temperature is important for high technology applications of polymers or, for example, in the food industry. The relaxation effect itself has been known since the work of Kohlrausch around 1876, then with renewed interest in 1930s [97] and came under research scrutiny about 50 years ago [32]. Presently, experimental methods for observing the physical relaxation of polymers are sufficiently large in number, and most efforts in this field are directed to studying changes in volume or enthalpy [227, 228]. The subject matter of this section includes the simulation of experimental data and research prospects in this field.

Early work on relaxation was focused on the temperature range $T \ge T_g$, because the relaxation time of the system at these temperatures is sufficiently short (on the order of seconds). The temperature dependence of the relaxation time is determined either by the Arrhenius law or by the VFT equation, depending on the system. The experimental data demonstrate well the nonlinear and nonexponential nature of glass transition relaxation. Naturally, we need methods ensuring corresponding properties to be able to conduct simulation studies.

A number of simple but adequate semiempirical expressions for the dependence of the integral characteristic curves C_p (for example, for the enthalpy) on relaxation temperature and time can be found in Refs [162, 229, 230]. These references allow the conclusion that the structure of a polymer affects the relaxation process. At the same time, the full description of the curves $C_p(T)$ requires a different approach. The application in this case of TNM or similar methods suggests that the experimental relaxation ('aging') interval of the polymer is divided into intervals, whose number is taken to be large enough to secure the required simulation precision. The simulation is carried out for heat capacity curves under heating, which are subsequently compared with their experimental counterparts (Fig. 10). These curves are then either considered directly as modeling the experimental data or used to obtain integral characteristics whose dependence on the relaxation temperature and time is being investigated.

Overall, there are three stages in the simulation of experiment: (1) initially, the liquid is cooled to a specified temperature T_a at a specified rate; (2) the liquid relaxes for a specified period of time t_a , and (3) the liquid is heated at a fixed rate q. The measurement and simulation of the relaxation kinetics in various polymers are detailed in a series of references [57, 58, 129, 231, 232] (see also review [61]). Some of the model parameters (for example, Δh^*) were determined from the characteristic behavior of measured quantities, and others by a least-mean-square fit of the model results to the experimental data. It is shown that the TNM and AG models reproduce rather well the heat capacity curves of polymers heated following relaxation; on the other hand, the parameters of the method are found to be manifestly dependent on the thermal history of the experiment—in particular, the parameters x and β should be varied when T_a is changed by 10 K or more or when t_a is changed by an order of magnitude or more.

Detailed studies of the relaxation of polyetherimide were performed in Refs [233, 234]. In Refs [59, 159], data on polystyrene and three of its derivatives are described within the model of limiting configurational entropy. That the data turn out to be fitted better than by TNM is expected due to the added parameter. Importantly, even varying T_a by 20 K and t_a between 300 and 1000 min did not produce full agreement with experimental data. A subsequent study [158] on polystyrene-based mixtures estimated the size of inhomogeneities as a function of the mixture composition. Reference [235] compared the relaxation kinetics of the volume and entropy of polystyrene at temperatures close to T_g . In the neighbor-

Figure 10. (Color online.) Example of heat capacity measurements in polystyrene during heating following relaxation at different time intervals [65].



hood of the glass transition temperature, their behavior is the same, but extrapolation to the region $T_g - 20$ K should lead to differences according to the authors. At the same time, similar experiments in Ref. [236] show that the relaxation of volume and enthalpy are of the same nature and have the same parameters.

Over the last five years, most discussion in this area has been concerned with the relaxation mechanism we described in the Introduction, a two-stage mechanism, so called due to the intermediate state ('plateau') present between the glassy and liquid states (see Fig. 3). In some early studies [59, 230], similar hypotheses were suggested based both on experimental data and simulation results, but it is only in 2011 that such observations were made directly in an experiment [63]. In subsequent work, some groups refuted [64], whereas others [65] improved-and hence confirmed-these results. It should be noted that these results were obtained by only one group of researchers [63, 65, 227] and have not yet been confirmed independently. Because a single experiment in this field takes over a year to complete, it is expected that in the next one or two years new data will be obtained which will enable a more definitive conclusion to be drawn. Anyway, this research holds great promise for the development of the theory of the glass transition, as does the simulation of the two-stage relaxation process from the fundamental standpoint.

6. Glass transition kinetics in a wide range of cooling rates

Along with the study of physical relaxation, the kinetics of the polymer glass transition under linear variation of temperature over a given range are being investigated, and a rather large amount of literature has accumulated. Zhurkov and Levin [109] were the first to observe a peak in the temperature dependence of the heat capacity of polyvinyl acetate and polymethyl methacrylate at two different heating rates. The dependence of the glass transition temperature as described by Bartenev [21] is also well manifested in polymers [21, 109, 113]. Subsequent decades have witnessed the gradual extension of the range of accessible cooling rates and an increase in measurement precision. As new theories of glass transition kinetics appear, their potential in relation to experimental data is continuously examined [57, 147, 151, 232, 237–243].

Recent years have seen a renewal of activity in this field due, in particular, to the ability of modern DCS instruments to vary the cooling rate over a very wide range. As a result, the existing glass transition models have been verified and alternative approaches proposed. Measurements of the dependence of the glass transition temperature of polystyrene on the cooling rate are presented in Refs [244, 245]. Reference [246] investigated the glass transition of polystyrene over the widest cooling rate range ever reported $(5 \times 10^{-6}-2 \text{ K s}^{-1})$.

Figure 11 demonstrates the characteristic experimental curves and simulation results in the form of the values of the heat capacity C_p^{norm} equal to the difference of two reduced heat capacities, one of which is measured under heating at a rate of 0.5 K s⁻¹ following cooling at an indicated rate taken from the given interval, and the other at heating at 0.5 K s⁻¹ following cooling at 0.5 K s⁻¹ (this measurement scenario is needed to improve the quality of the resulting curves at sufficiently slow cooling). Modeling the full set of curves first using the TNM and AG models and then in the



Figure 11. (Color online.) Temperature dependence of the heat capacity of polystyrene for different cooling rates as measured under heating at a rate of 0.5 K s⁻¹ and the model data [246]. Thick lines — experiment, and thin lines — Adam–Gibbs simulation.

framework of nonequilibrium thermodynamics (see papers [174, 177]) with the same parameter set yields qualitatively similar results (see Fig. 11).

To perform the TNM simulation, expression (18) is modified by noting that polystyrene belongs to the class of hard glass-forming systems, so one has

$$\tau(T, T_{\rm f}) = \exp\left[\ln\tau_0 + \frac{\Delta h^*}{R(T - T_0)} x + \frac{\Delta h^*}{R(T_{\rm f} - T_0)} (1 - x)\right].$$
(64)

The Adam–Gibbs theoretical approach implies the specification of the dependence $\Delta C_p(T)$ for the system under study. Review [61] suggests a number of expressions (a constant, and linear and hyperbolic dependences), each of which was used to calculate the model curves in Ref. [246]. The best agreement with experiment was achieved with the linear dependence $\Delta C_p(T) = C_0 + C_1 T$, which, when substituted into Eqns (23) and (25), yields

$$\tau_{\rm AG}(T, T_{\rm f}) = \tau_0 \exp\left\{\frac{A}{RT \left[\ln\left(T_{\rm f}/T_2\right) + C(T_{\rm f} - T_2)\right]}\right\}.$$
 (65)

Finally, while a number of expressions for τ were tested for the purpose of a description in the framework of nonequilibrium thermodynamics (see Section 3.8), the best agreement with experiment was obtained with a new expression, which combines the VFT law and formula (23) from the AG theory:

$$\tau(\theta,\xi) = \tau_0 \exp\left\{\frac{B}{R(\theta-\theta_0)} - \frac{A}{R\theta\left[\ln\left(1-\xi\right) + \xi\ln\xi/(1-\xi)\right]}\right\},\tag{66}$$

where θ is the reduced temperature, the first term in the exponential is the VFT law, and the second term is determined, according to the AG theory, by the configurational part of the entropy of the system.

It can be concluded that, given such a wide range of heating/cooling rates q, none of the methods can provide a sufficient accuracy in describing experimental data. A point to note in Fig. 11 is that discrepancies occur not only in the



Figure 12. (Color online.) TNM simulation of isobaric heat capacity curves by varying the Δh and β parameters for each measurement (heating or cooling) [246]. Thick lines — experiment, and thin lines — Narayanaswamy's simulation.

high C_p peaks corresponding to slow cooling and, hence, to the parallel existence of structural relaxation, but also at relatively large rates, for which lower-than-measured heat capacity peaks are predicted.

Reference [246] argues, consistent with previous understanding, that the exact simulation of the curves can be achieved only by varying a certain two of the method's parameters (for example, Δh and β) at a variable rate q. That is, if for each curve $C_p(T)$ —and, notably, for heating and cooling separately, because the absolute values of the rates are different—one automatically chooses two parameters and leaves the others fixed, then a qualitative agreement between the theoretical and experimental curves can be achieved for most of the methods discussed in Section 2. Results of such calculations, conducted by twenty experimental curves (nineteen for cooling and one for heating) are presented in Fig. 12. The small shape differences between the model curves (especially near the minimum of C_p) are constant for different values of the cooling rate q and, hence, can be viewed as a drawback of the simulation method or as a specific feature of the glass transition in polystyrene.

There are two factors qualitatively supplementing the analysis of the fit between the model and measured curves, namely, the descriptive behavior characteristics of the curve parameters such as, first, the position and value of the C_p maximum and, second, their integral characteristics, the limiting fictive temperature T'_f proportional to the integral of the enthalpy H:

$$T_{\rm f}' = T_{\rm f}^0 - \frac{\Delta H}{\Delta C_p} \,, \tag{67}$$

where T_f^0 is the reference value of the fictive temperature determined by Eqn (6) from an individual measurement of C_p under heating and cooling at the same rate of 0.5 K s⁻¹ [246]. It is shown that the TNM model yields better agreement with the experiment for the positions of the heat capacity peaks, whereas the AG method performs better for the values of C_p peaks.

Figure 13 compares the dependence of the limiting fictive temperature on the cooling rate q for different simulation methods. It is seen that when using a single set



Figure 13. Comparison of the experimental (squares) and calculated values of the limiting fictive temperature. TNM for constant parameters — diamonds; AG for constant parameters — dots; TNM for varying parameters — triangles.

of simulation parameters, the slope of the curve $T'_f(q)$ is substantially different from what is measured, whereas varying two parameters gives a sufficiently good fit. Also note that the model curves intersect the experimental one in the vicinity of q = 0.5 K s⁻¹, which corresponds to an experiment with equal heating and cooling rates. The implication of this is that the quality of the glass transition kinetics simulation is determined by the heating-to-cooling rate ratio and worsens when this ratio is significantly different from unity [242].

Reference [246] also examined whether the linear extrapolation of the given dependences (Δh and β on q) can be used to predict experimental results. It is shown that the parameters that describe C_p curves best do not correspond to the qualitative description of the dependence of the integral characteristics and the fictive temperature on q.

Presently, similar experimental data are available model simulations are being carried out for polystyrene at large cooling rates (the cooling rate range is $0.1-10^4$ K s⁻¹, the corresponding heating rate is 5000 K s⁻¹), the modeling of the results is taking place. It is shown that the results of Ref. [246] are clearly manifested for the simultaneous simulation of two sets of data; theoretical methods with a constant set of parameters are unable to describe the observed patterns. It should be noted that other approaches—for example, the configurational entropy hypothesis-run into the same difficulties. For example, in the work carried out in Refs [119, 164] on the polymer glass transition in the range of rates q up to three orders, only the introduction of new phenomenological laws into the dependences of the parameters on the cooling rate was able to improve the agreement between the theoretical and experimental results.

7. Conclusions

Research on and the use of the glass transition long ago entered the sphere of human activity. Studying and optimizing methods for fabricating and processing glass were the preoccupations of people as far back as about five millennia ago. Evidence of interest in glass in ancient civilizations has survived until the present day.

Currently, glass transition kinetics and related processes are one of the central problems in modern condensed matter physics, with original and review papers on the subject numbering in the tens annually. The glass transition process, as noted in the Introduction, is very general in character, similar processes occurring in systems totally different in terms of their properties and the interactions involved. Thus, of extreme interest are spin, structural (covalent, metallic, polymer), dipole, and superconducting glasses. A relatively new class of materials are polymer-based nanocomposite materials [227, 247–249] with various nanoparticle inclusions, among them new promising allotropic carbon forms such as fullerenes and carbon nanotubes [250-254]. The systems listed either are already used in everyday life or hold vast promise for high-technology applications. In this connection, controling the glass transition process and material properties and predicting the dynamics of property changes under working and extreme temperatures are problems of great importance for practical applications.

Over a few decades, there has been an evolution in describing the glass transition, from simple phenomenological first-order differential equations to modern methods of nonequilibrium thermodynamics and statistical physics. Increasingly, the computer simulation of systems with simple interactions is gaining importance [255–259], which enables the current microscopic theories of the glass transition to be tested. At the same time, as can also be seen from the present review, we are still far from having a complete theory of the glass transition capable of describing the evolution of the full set of properties of a system at temperatures from $T_{\rm m}$ to $T_{\rm g}$ (and lower) for an arbitrary experimental thermal history and to fit the wide range of currently available experimental data.

Microscopic approaches and computer simulation deal well with describing certain patterns in the temperature range from the melting point to the glass transition temperature and in some cases are capable of making qualitative predictions. However, close to and within the glass transition range certain corrections should be introduced to such methods as the mode-coupling theory or the random first-order transition theory. Computer simulation does not yet have sufficient computational power to investigate the behavior of systems close to actual ones (at actual cooling rates) in the region of the glass transition, where relaxation lasts for tens or hundreds of seconds. In our view, what the glass transition kinetics of various materials require for its description are approaches based on the thermodynamics of irreversible processes and combined with accurately selected statistical models grounded as far as possible on microscopic considerations. As the glass transition theory develops further, this approach should allow a sufficiently accurate description of experimental data over a wide range of temperatures and of the rates of change of external parameters.

This review presents the state of the art in the study of theoretical approaches to describing glass transition kinetics and physical relaxation. It considers classical theories and a number of present-day methods, analyzes the published studies, and generalizes their results with emphasis on the investigation of the temperature evolution of the heat capacity of polymers in the glass transition interval. This choice, as can be seen, does not narrow the range of applicability of the methods described—indeed, they are adequate for virtually any problem related to the description of the glass transition. Moreover, part of the presented theories have their origin in other research fields such as spin systems.

The analysis of a wide range of methods and studies which is presented here suggests that at present most calorimetry experiments are described only in the framework of phenomenological approaches such as the Narayanaswamy method or the Adam–Gibbs theory. Furthermore, a detailed description of experiments requires varying model parameters, taking into account their estimates from experiments.

As a promising avenue in the development of theoretical methods in future work, we note the development of microscopic theories of the process and the transition from the description and computer simulation of simple classical experiments to the computer simulation of more complex experiments. Modern theories offer the possibility of the nonempirical description of the general features of the glass transition; in particular, success was achieved in describing the Vogel–Fulcher–Tammann and Kohlrausch laws. The advantages of the modern theories also include the possibility of obtaining predictions by initially specifying the exact expressions for the interaction between the system's particles. Modern computer centers have enough power to perform some calculations in the glass transition temperature range for molecular systems [260] and polymers [261].

With modern experimental techniques, the external parameters and the thermal history characteristics can be varied by more than ten orders of magnitude. For example, in the experiments of Ref. [246], the cooling rate varied by six orders, and adding data from fast calorimetry will extend the range of the parameter q by up to 10 orders. Similarly, studies of the phenomenon of dynamic glass transition [42] allow the frequency of the external influence to be varied over a wide range of $10^{-5}-10^{6}$ Hz.

However, none of the theoretical approaches listed in this review is yet capable of coping with the problem of the detailed description of experimental data. Another problem that also remains to be solved is the study of the physical relaxation of supercooled systems [65]; at issue here is whether the observed relaxation mechanism is evidence for the bimodal distribution of relaxation times [223] or whether it provides a direct indication of the existence of an 'intermediate plateau' in the relaxation process [59]. Solving this and other complex problems requires the development of existing glass transition theories. The possible next step on this path is the development of new physically reasonable expressions for the relaxation time or allowing switching between relaxation mechanisms in a certain temperature range in the glass transition interval or above. It would be a breakthrough if an unequivocal, logically grounded experiment was designed to answer the question of whether the glass transition is a purely kinetic phenomenon or a consequence of the phase transition to the 'ideal glass' state.

As regards nonphenomenological theories, such as the mode-coupling theory, these should be developed to describe a great variety of accumulated experimental data. It should be noted that most theoretical studies in this field usually discuss only those experimental data that support the theory or else examine simulation with a simple interaction. Heat capacity curves, for example, do not belong to this category of data, so a necessary step in the development of the theory of the glass transition would be to propose semiempirical expressions capable of describing the $C_p(T)$ dependences at a level not inferior to the existing phenomenological approaches.

Physics-Uspekhi 59 (1)

Thus, describing the glass transition phenomenon remains a problem of great importance for virtually all fundamental and application fields, and investigations in this area are of extreme interest.

Acknowledgments

64

Financial support from the German Federal Ministry of Education and Scientific Research (BMBF) within the Heisenberg-Landau program is gratefully acknowledged.

References

- 1. Schmelzer J W P et al. Glasses and the Glass Transition (Weinheim: Wiley-VCH, 2011)
- 2 Mazurin O V Steklovanie (Glass Transition) (Leningrad: Nauka, 1986)
- 3. Gutzow I, Schmelzer J The Vitreous State: Thermodynamics, Structure, Rheology, and Crystallization (Berlin: Springer, 1995)
- Gutzow I S, Schmelzer J W P The Vitreous State: Thermodynamics, 4. Structure, Rheology, and Crystallization 2nd ed. (Berlin: Springer, 2013)
- 5. Aksenov V L, Bobeth M, Schreiber J Theor. Math. Phys. 64 770 (1985); Teor. Mat. Fiz. 64 186 (1985)
- Aksenov V L, in V Mezhdunar. Shkola po Neitronnoi Fizike, Alushta, 6. 1986. Sbornik Lektsii (Vth Intern. School on Neutron Physics, Alushta, 1986. Lectures) (Dubna: JINR, 1986) p. 273
- 7. Müller K, Takashige M, Bednorz J Phys. Rev. Lett. 58 1143 (1987)
- Aksenov V L, Sergeenkov S A Physica C 156 235 (1988) 8.
- 9. Aksenov V L Ordering Phenomena in Condensed Matter Physics (Teaneck, NJ: World Scientific, 1991) pp. 2-23
- 10. Andrievski R A Phys. Usp. 56 261 (2013); Usp. Fiz. Nauk 183 277 (2013)
- Binder K, Young A P Rev. Mod. Phys. 58 801 (1986) 11.
- 12. Cangialosi D J. Phys. Condens. Matter 26 153101 (2014)
- 13. Ngai K L Relaxation and Diffusion in Complex Systems (New York: Springer, 2011)
- Schmelzer J W P, Tropin T V J. Non-Cryst. Solids 407 170 (2015) 14.
- Kohlrausch R Ann. Physik 91 179 (1854) 15.
- 16. Williams G, Watts D C Trans. Faraday Soc. 66 80 (1970)
- Tropin T V, Schmelzer J W P, Schick C J. Non-Cryst. Solids 357 17. 1291 (2011)
- 18 Tropin T V, Schmelzer J W P, Schick C J. Non-Cryst. Solids 357 1303 (2011)
- 19. Schmelzer J W P J. Chem. Phys. 136 074512 (2012)
- Bartenev G M Dokl. Akad. Nauk SSSR 76 227 (1951) 20.
- Bartenev G M, Luk'yanov I A Zh. Fiz. Khim. 29 1486 (1955) 21.
- Ritland H N J. Am. Ceram. Soc. 37 370 (1954) 22.
- 23. Angell C A Science 267 1924 (1995)
- Tammann G Der Glaszustand (Leipzig: L. Voss, 1933) 24
- 25 Schmelzer J W P et al. J. Non-Cryst. Solids (2015)
- 26 Vogel H Phys. Z. 22 645 (1921)
- Fulcher G S J. Am. Ceram. Soc. 8 339 (1925) 27.
- Tammann G, Hesse W Z. Anorg. Allg. Chem. 156 245 (1926) 28.
- Ritland H N J. Am. Ceram. Soc. 39 403 (1956) 29.
- Spinner S, Napolitano A J. Res. Natl. Bur. Stand. A 70 147 (1966) 30.
- Boesch L, Napolitano A, Macedo P B J. Am. Ceram. Soc. 53 148 31.
- (1970)32 Kovacs A J Adv. Polym. Sci. 3 394 (1964)
- Miyamoto Y et al. Phys. Rev. Lett. 88 225504 (2002) 33
- Chen W et al. Front. Chem. China 4 229 (2009) 34.
- Allen L H et al. Appl. Phys. Lett. 64 417 (1994) 35.
- Minakov A A, Schick C Rev. Sci. Instrum. 78 (2007) 36.
- Šesták J, Holba P J. Therm. Anal. Calorim. (2013) 37.
- 38. Birge N, Nagel S Phys. Rev. Lett. 54 2674 (1985)
- 39 Birge N Phys. Rev. B 34 1631 (1986)
- 40. Merzlyakov M, Schick C Thermochim. Acta 330 55 (1999)
- Merzlyakov M, Schick C Thermochim. Acta 330 65 (1999) 41.
- Chua Y Z et al. Colloid Polym. Sci. 292 1893 (2014) 42.
- Huth H et al. Eur. Phys. J. Spec. Top. 141 153 (2007) 43.
- Weyer S, Merzlyakov M, Schick C Thermochim. Acta 377 85 (2001) 44.
- 45. Weyer S, Huth H, Schick C Polymer 46 12240 (2005)

- Weyer S et al. Thermochim. Acta 304-305 251 (1997) 46.
- 47. Hempel E et al. J. Phys. Chem. B 104 2460 (2000)
- 48. Donth E Relaxation and Thermodynamics in Polymer-Glass Transition (Berlin: Akademie Verlag, 1992)
- 49 Hensel A, Schick C J. Non-Cryst. Solids 235-237 510 (1998)
- 50 Huth H et al. Thermochim. Acta 377 113 (2001)
- Schick C, in Handbook of Thermal Analysis and Calorimetry: 51. Applications to Polymers and Plastics (Eds P K Gallagher, S Z D Cheng) Vol. 3 (Amsterdam: Elsevier, 2002) p. 713
- 52. Herzfeld K, Rice F Phys. Rev. 31 691 (1928)
- 53 Jeong Y Thermochim. Acta 304-305 67 (1997)
- 54. Alig I Thermochim. Acta 304-305 35 (1997)
- 55 Schawe J E K Thermochim. Acta 261 183 (1995)
- 56 Chen H S, Wang T T J. Appl. Phys. 52 5898 (1981)
- Hodge I M, Huvard G S Macromolecules 16 371 (1983) 57.
- Hodge I M Macromolecules 16 898 (1983) 58.
- 59. Brunacci A et al. Macromolecules 29 7976 (1996)
- Hutchinson J M, Kumar P Thermochim. Acta 391 197 (2002) 60.
- 61. Hodge I M J. Non-Cryst. Solids 169 211 (1994)
- Sakatsuji W, Konishi T, Miyamoto Y J. Therm. Anal. Calorim. 113 62. 1129 (2013)
- 63. Boucher V M et al. Macromolecules 44 8333 (2011)
- Koh Y P, Simon S L Macromolecules 46 5815 (2013) 64
- 65. Cangialosi D et al. Phys. Rev. Lett. 111 095701 (2013)
- Gómez Ribelles J L et al. Polymer 38 963 (1997) 66.
- Schick C Eur. Phys. J. Spec. Top. 189 3 (2010) 67.
- 68. Forrest J A, Dalnoki-Veress K Adv. Colloid. 94 167 (2001)
- 69. Priestley R D, Cangialosi D, Napolitano S J. Non-Cryst. Solids 407 288 (2014)
- Koh Y P, Grassia L, Simon S L Thermochim. Acta 603 135 (2015) 70
- 71. Boucher V M et al. Thermochim. Acta 575 233 (2014)
- Keddie J L, Jones R A L, Cory R A Europhys. Lett. 27 59 (2007) 72.
- Prucker O et al. Macromol. Chem. Phys. 199 1435 (1998) 73.
- 74. Kawana S, Jones R A L Phys. Rev. E 63 021501 (2001)
- 75. Keddie J L, Jones R A L, Cory R A Faraday Discuss. 98 219 (1994)
- 76. Grohens Y et al. Langmuir 14 2929 (1998)
- 77. Mattsson J, Forrest J A, Börjesson L Phys. Rev. E 62 5187 (2000)
- 78. Sharp J S, Forrest J A Phys. Rev. E 67 031805 (2003)
- 79 Svanberg C Macromolecules 40 312 (2007)
- 80 Boucher V M et al. Macromolecules 45 5296 (2012)
- Mazurin O V J. Non-Cryst. Solids 351 1103 (2005) 81.
- 82. Wunderlich B Pure Appl. Chem. 67 1019 (1995)
- Pyda M Thermal Properties of Polymers (Springer Materials) (New 83. York: Springer, 2013)
- 84. Kauzmann W Chem. Rev. 43 219 (1948)
- Mazurin O V J. Non-Cryst. Solids 25 129 (1977) 85
- Mazurin O V J. Non-Cryst. Solids 87 392 (1986) 86.
- 87 Ediger M D, Angell C A, Nagel S R J. Phys. Chem. 100 13200 (1996)
- 88. Angell C A Curr. Opin. Solid State Mater. Sci. 1 578 (1996)
- 89. Angell C A et al. J. Appl. Phys. 88 3113 (2000)
- 90. Debenedetti P G, Stillinger F H Nature 410 259 (2001)
- 91. Stillinger F H, Debenedetti P G Annu. Rev. Condens. Matter Phys. 4 263 (2013)

Gutzow I, Schmelzer J W P, Petroff B J. Non-Cryst. Solids 354 311

- 92. Biroli G, Garrahan J P J. Chem. Phys. 138 12A301 (2013)
- 93 Cavagna A Phys. Rep. 476 51 (2009)
- 94. McKenna G B Nature Phys. 4 673 (2008) Pye L D, Montenero A, Joseph I Properties of Glass-Forming Melts

(Boca Raton: Taylor and Francis, 2005)

Simon F Trans. Faraday Soc. 33 65 (1937)

104. Davies R O, Jones G O Adv. Phys. 2 370 (1953)

100. Tool A Q J. Am. Ceram. Soc. 29 240 (1946)

101. Tool A Q J. Am. Ceram. Soc. 31 177 (1948)

Publ. Department, 2008) p. 303

Simon F Z. Anorg. Allg. Chemie 203 219 (1931)

Tool A Q, Eichlin C G J. Am. Ceram. Soc. 14 276 (1931)

Badrinarayanan P et al. J. Non-Cryst. Solids 353 2603 (2007)

105. Schmelzer J W P, Gutzow I S, in Nucleation Theory and Applications

(Eds J W P Schmelzer, G Röpke, V B Priezzhev) (Dubna: JINR

Moynihan C T et al. J. Am. Ceram. Soc. 59 12 (1976)

106. Mauro J C, Loucks R J J. Non-Cryst. Solids 355 676 (2009)

95.

96.

97.

98

99.

102.

103.

(2008)

- Vol'kenshtein M V, Ptitsyn O B Dokl. Akad. Nauk SSSR 103 795 (1955)
- 108. Vol'kenshtein M V, Ptitsyn O B Zh. Tekh. Fiz. 26 2204 (1956)
- 109. Zhurkov S N, Levin B Ya, in Sbornik, Posvyashchennyi Semidesyatiletiyu Akademika A.F. Ioffe (Collection Honoring the 70th Anniversary of the Birth of Academician A.F. Ioffe) (Moscow-Leningrad: Izd. AN SSSR, 1950) p. 260
- 110. Cooper A R, Gupta P K Phys. Chem. Glas. 23 44 (1982)
- 111. Doolittle A K J. Appl. Phys. 22 1471 (1951)
- 112. Turnbull D, Cohen M H J. Chem. Phys. 29 1049 (1958)
- 113. Fox T G, Flory P J J. Polym. Sci. 14 315 (1954)
- Frenkel J Kinetic Theory of Liquids (Oxford: The Clarendon Press, 1946); Translated from Russian: Kineticheskaya Teoriya Zhidkostei (Leningrad: Nauka, 1975)
- Grest G S, Cohen M H, in Advances in Chemical Physics Vol. 48 (Eds I Prigogine, S A Rice) (New York: Wiley, 1981) p. 455
- 116. Macedo P B, Litovitz T A J. Chem. Phys. 42 245 (1965)
- 117. Hutchinson J M, Kovacs A J J. Polym. Sci. Polym. Phys. Ed. 141575 (1976)
- 118. Cohen M, Grest G Phys. Rev. B 20 1077 (1979)
- 119. Liu G, Zhao D, Zuo Y J. Non-Cryst. Solids 417-418 52 (2015)
- 120. Narayanaswamy O S J. Am. Ceram. Soc. 54 491 (1971)
- 121. Moynihan C T et al. Ann. New York Acad. Sci. 279 15 (1976)
- 122. DeBolt M A et al. J. Am. Ceram. Soc. 59 16 (1976)
- 123. Naravanaswamy O S J. Am. Ceram. Soc. 64 109 (1981)
- 124. Gardon R, Narayanaswamy O S J. Am. Ceram. Soc. 53 380 (1970)
- 125. Narayanaswamy O S J. Am. Ceram. Soc. 60 1 (1977)
- 126. Narayanaswamy O S J. Am. Ceram. Soc. 61 146 (1978)
- 127. Narayanaswamy O S J. Am. Ceram. Soc. 71 900 (1988)
- Moynihan C T, Crichton S N, Opalka S M J. Non-Cryst. Solids 131-133 420 (1991)
- 129. Hodge I M, Berens A R Macromolecules 15 762 (1982)
- 130. Simon S L, McKenna G B Thermochim. Acta 307 1 (1997)
- 131. Simon S L, McKenna G B J. Chem. Phys. 107 8678 (1997)
- 132. Flikkema E Macromolecules 31 892 (1998)
- 133. Simon S L, McKenna G B Thermochim. Acta 348 77 (2000)
- 134. Mazurin O V, Rekhson S M, Startsev Yu K Fiz. Khim. Stekla 1 438 (1975)
- 135. Mazurin O V, Startsev Yu K Fiz. Khim. Stekla 7 408 (1981)
- 136. Kovacs A J et al. J. Polym. Sci. Polym. Phys. Ed. 17 1097 (1979)
- 137. Ramos A R et al. J. Polym. Sci. Part B Polym. Phys. 26 501 (1988)
- 138. Hutchinson J M, Kovacs A J Polym. Eng. Sci. 24 1087 (1984)
- 139. Richert R Phys. Rev. Lett. 104 249801 (2010)
- 140. Richert R J. Chem. Phys. 134 144501 (2011)
- 141. Mazinani S K S, Richert R J. Chem. Phys. 136 174515 (2012)
- 142. Giordano M et al. J. Non-Cryst. Solids **351** 515 (2005)
- 143. Hutchinson J M, Montserrat S J. Therm. Anal. 47 103 (1996)
- 144. Hutchinson J M, Montserrat S Thermochim. Acta 377 63 (2001)
- 145. Adam G, Gibbs J H J. Chem. Phys. 43 139 (1965)
- 146. Gibbs J H, DiMarzio E A J. Chem. Phys. 28 373 (1958)
- 147. Kusy R P, Greenberg A R, Hill C Polymer 23 36 (1982)
- 148. Scherer G W J. Am. Ceram. Soc. 67 504 (1984)
- 149. Hodge I M J. Non-Cryst. Solids 131-133 435 (1991)
- 150. Andreozzi L et al. J. Phys. Condens. Matter 15 S1215 (2003)
- 151. Vyazovkin S, Dranca I J. Phys. Chem. B 108 11981 (2004)
- 152. Liu G et al. Eur. Phys. J. E 37 63 (2014)
- 153. Odagaki T et al. J. Non-Cryst. Solids 352 4843 (2006)
- 154. Drozdov A D Polym. Bull. 45 303 (2000)
- 155. Ngai K L J. Phys. Chem. B 103 5895 (1999)
- Gómez Ribelles J L, Monleón Pradas M Macromolecules 28 5867 (1995)
- 157. Gómez Ribelles J L et al. Macromolecules 28 5878 (1995)
- 158. Cowie J M G et al. *Macromolecules* **32** 4430 (1999)
- 159. Salmerón M et al. Colloid Polym. Sci. 277 1033 (1999)
- 160. Alves N M et al. *Polymer* **43** 4111 (2002)
- 161. Mano J F et al. Polymer 46 8258 (2005)
- 162. Cowie J, Ferguson R Macromolecules 2307 (1989)
- 163. Ribelles J L G et al. *Polymer* **28** 2262 (1987)
- 164. Liu G et al. J. Non-Cryst. Solids 402 160 (2014)
- 165. Hutchinson J M et al. Macromolecules 33 5252 (2000)
- 166. Andreozzi L et al. Eur. Phys. J. B 41 383 (2004)
- De Donder T, van Rysselberghe P Thermodynamic Theory of Affinity (London: Stanford Univ. Press, 1936)

 Prigogine I, Defay R Chemical Thermodynamics (London: Longmans, 1954)

65

- 169. Gutzow I et al. J. Chem. Phys. 112 10941 (2000)
- 170. Gutzow I, Dobreva A Polymer 33 451 (1992)
- 171. Nemilov S V Thermodynamic and Kinetic Aspects of the Vitreous State (Boca Raton: CRC Press, 1995)
- 172. Bragg W L, Williams E J Proc. R. Soc. Lond. A 145 699 (1934)
- 173. Mandelstam L I, Leontovich M A Zh. Eksp. Teor. Fiz. 7 438 (1937)
- 174. Möller J, Gutzow I, Schmelzer J W P J. Chem. Phys. 125 094505 (2006)
- 175. Gutzow I, Grigorova T, Todorova S J. Non-Cryst. Solids 304 4 (2002)
- 176. Moynihan C T et al. J. Phys. Chem. 78 2673 (1974)
- 177. Schmelzer J W P, Tropin T V J. Chem. Phys. 138 034507 (2013)
- 178. Buchenau U Phys. Rev. B 86 184105 (2012)
- 179. Tropin T V et al. J. Chem. Phys. 136 124502 (2012)
- 180. Goldstein M J. Chem. Phys. 51 3728 (1969)
- 181. Stillinger F H, Weber T A Phys. Rev. A 25 978 (1982)
- 182. Stillinger F H, Weber T A Phys. Rev. A 28 2408 (1983)
- 183. Angell C A J. Therm. Anal. Calorim. 69 785 (2002)
- 184. Odagaki T, Yoshimori A J. Non-Cryst. Solids 355 681 (2009)
- 185. Odagaki T, Ekimoto T J. Non-Cryst. Solids 353 3928 (2007)
- 186. Yoshimori A, Odagaki T J. Phys. Soc. Jpn. 74 1206 (2005)
- 187. Odagaki T, Hiwatari Y Phys. Rev. A 41 929 (1990)
- 188. Odagaki T Phys. Rev. Lett. 75 3701 (1995)
- 189. Odagaki T Prog. Theor. Phys. Suppl. 126 9 (1997)
- Tao T, Yoshimori A, Odagaki T *Phys. Rev. E* 64 046112 (2001)
- Tao T, Yoshimori A, Odagaki T Phys. Rev. E 66 041103 (2002)
 Tao T, Yoshimori A, Odagaki T Phys. Rev. E 66 041103 (2002)
- Ido J, Formini T, Sodagan T 195 Rev. 2 60 01100 (2002)
 Odagaki T, Tao T, Yoshimori A J. Non-Cryst. Solids 307 310 407 (2002)
- 193. Odagaki T J. Chem. Phys. 117 10151 (2002)
- 194. Tao T, Odagaki T, Yoshimori A J. Chem. Phys. 122 44505 (2005)
- 195. Sakatsuji W, Konishi T, Miyamoto Y Phys. Rev. E 88 012605 (2013)
- 196. Roland C M, Santangelo P G, Ngai K L J. Chem. Phys. 111 5593 (1999)
- 197. Mauro J C, Loucks R J, Gupta P K J. Phys. Chem. A 111 7957
- (2007) 198. Mauro J, Loucks R *Phys. Rev. B* **76** 174202 (2007)
- 199. Mauro J, Allan D, Potuzak M *Phys. Rev. B* **80** 094204 (2009)
- 1)). Maulo J, Alan D, I otuzak M I hys. Rev. D 60 094204 (200
- 200. Leutheusser E Phys. Rev. A 29 2765 (1984)

205. Das S Rev. Mod. Phys. 76 785 (2004)

216. Fuks N Usp. Fiz. Nauk 15 496 (1936)

P05013 (2005)

204.

206.

207.

208.

209.

212.

214

215.

217.

221.

222.

(1989)

 Bengtzelius U, Gotze W, Sjolander A J. Phys. C Solid State Phys. 17 5915 (2000)

Kob W, in Experimental and Theoretical Approaches to Supercooled

Liquids: Advances and Novel Applications (Eds J Fourkas et al.)

Reichman D R, Charbonneau P J. Stat. Mech. Theory Exp. 2005

Gotze W, in Liquids, Freezing and Glass Transition, Les Houches, Session LI, 1989 (Eds J P Hansen, D Levesque, J Zinn-Justin)

(Amsterdam: Elsevier, 1991); Translated into Russian: Fazovye

Kirkpatrick T R, Thirumalai D, Wolynes P G Phys. Rev. A 40 1045

Gross D J, Kanter I, Sompolinsky H Phys. Rev. Lett. 55 304 (1985)

Schmelzer J W P, in Encyclopedia of Surface and Colloid Science

(Ed. A T Hubbard) (New York: Marcel Dekker, 2002) p. 4017

211. Lubchenko V, Wolynes P G Annu. Rev. Phys. Chem. 58 235 (2007)

Kirkpatrick T R, Wolynes P G Phys. Rev. B 36 8552 (1987)

- 202. Osada H Probab. Theory Relat. Fields 112 53 (1998)
- 203. Schweizer K S, Saltzman E J J. Chem. Phys. 119 1181 (2003)

(Washington: ACS Books, 1997) p. 28; cond-mat/9702073

Berthier L, Biroli G Rev. Mod. Phys. 83 587 (2011)

Perekhody Zhidkost'-Steklo (Moscow: Nauka, 1992)

213. Kirkpatrick T R, Wolynes P G Phys. Rev. A 35 3072 (1987)

Zel'dovich Ya B Zh. Eksp. Teor. Fiz. 12 525 (1942)

219. Thurau C T, Ediger M D J. Chem. Phys. 116 9089 (2002)

Johari G P, Khouri J J. Chem. Phys. 137 104502 (2012)

223. Wisitsorasak A, Wolynes P G J. Phys. Chem. B 118 7835 (2014)

Khouri J, Johari G P J. Chem. Phys. 138 2013 (2013)

218. Ediger M D Annu. Rev. Phys. Chem. 51 99 (2000)

220. Richert R J. Phys. Condens. Matter 14 R703 (2002)

210. Kirkpatrick T R, Thirumalai D, arXiv:1401.2024

- 224. Mazurin O V Glass Phys. Chem. 33 22 (2007); Fiz. Khim. Stekla 33 34 (2007)
- 225. Moynihan C T J. Am. Ceram. Soc. 76 1081 (1993)
- 226. Kobeko P P Amorfnye Veshchestva (Amorphous Substances) (Moscow: Izd. AN SSSR, 1952)
- 227. Cangialosi D et al. Soft Matter 9 8619 (2013)
- 228. Mijović J et al. Polym. Eng. Sci. 34 381 (1994)
- 229. Brunacci A et al. Polymer 38 865 (1997)
- 230. Brunacci A et al. Polymer 38 3263 (1997)
- 231. Hodge I M Macromolecules 20 2897 (1987)
- 232. O'Reilly J M, Hodge I M J. Non-Cryst. Solids 131-133 451 (1991)
- 233. Simon S L Macromolecules 9297 4056 (1997)
- 234. Echeverria I et al. J. Polym. Sci. Part B 33 2457 (1995)
- 235. Simon S L, Sobieski J W, Plazek D J Polymer 42 2555 (2001)
- 236. Hadač J et al. J. Non-Cryst. Solids **353** 2681 (2007)
- Wunderlich B, Bodily D M, Kaplan M H J. Appl. Phys. 35 95 (1964)
 Privalko V P, Demchenko S S, Lipatov Y S Macromolecules 19 901
- (1986) 239. Viollaz P E, Alzamora S M, Nieto A B *J. Food Eng.* **56** 393 (2003)
- 240. Chang S J. Polym. Sci. Polym. Symp. 71 59 (2007)
- 241. Tombari E et al. *Phys. Rev. B* **78** 144203 (2008)
- 242. Liu G et al. J. Non-Cryst. Solids 365 13 (2013)
- 243. Noda I, Dowrey A E, Marcott C Vib. Spectrosc. 51 22 (2009)
- 244. Schawe J E K J. Chem. Phys. 141 184905 (2014)
- 245. Schawe J E K Thermochim. Acta 603 128 (2015)
- 246. Tropin T V et al. J. Non-Cryst. Solids 409 63 (2015)
- 247. Balazs A C, Emrick T, Russell T P Science 314 1107 (2006)
- 248. Starr F W, Schrøder T B, Glotzer S C Macromolecules 35 4481 (2002)
- 249. Tuteja A, Duxbury P M, Mackay M E Macromolecules 40 9427 (2007)
- 250. Chua Y-C et al. Macromolecules 43 9578 (2010)
- 251. Wong H C et al. J. Mol. Liq. 153 79 (2010)
- 252. Sanz A et al. Polymer 68 47 (2015)
- 253. Kropka J M et al. Macromolecules 40 5424 (2007)
- 254. Cipriano B H et al. Polymer 49 4846 (2008)
- 255. Hoang V Van, Odagaki T J. Phys. Chem. B 115 6946 (2011)
- 256. Barrat J L, Klein M L Annu. Rev. Phys. Chem. 42 23 (1991)
- 257. Binder K, Baschnagel J, Paul W Prog. Polymer Sci. 28 115 (2003)
- 258. Hansen J-P Physica A 201 138 (1993)
- 259. Gordon J M, Gibbs J H, Fleming P D J. Chem. Phys. 65 2771 (1976)
- Odinokov A, Freidzon A, Bagaturyants A Chem. Phys. Lett. 633 41 (2015)
- 261. Sarangapani R, Reddy ST, Sikder A K J. Mol. Graph. Model. 57 114 (2015)